



The University of Manchester Research

In-person interventions to reduce social isolation and loneliness: An evidence and gap map

DOI: 10.1002/cl2.1408

Document Version

Final published version

Link to publication record in Manchester Research Explorer

Citation for published version (APA):

Welch, V., Ghogomu, E. T., Dowling, S., Barbeau, V. I., Al-zubaidi, A. A. A., Beveridge, E., Bondok, M., Desai, P., Doyle, R., Huang, J., Hussain, T., Jearvis, A., Jahel, F., Madani, L., Choo, W. Y., Yunus, R. M., Tengku mohd, T. A. M., Wadhwani, A., Ameer, A. A., ... Mikton, C. (2024). In-person interventions to reduce social isolation and loneliness: An evidence and gap map. *Campbell Systematic Reviews*, *20*(2). https://doi.org/10.1002/cl2.1408

Published in:

Campbell Systematic Reviews

Citing this paper

Please note that where the full-text provided on Manchester Research Explorer is the Author Accepted Manuscript or Proof version this may differ from the final Published version. If citing, it is advised that you check and use the publisher's definitive version.

General rights

Copyright and moral rights for the publications made accessible in the Research Explorer are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Takedown policy

If you believe that this document breaches copyright please refer to the University of Manchester's Takedown Procedures [http://man.ac.uk/04Y6Bo] or contact uml.scholarlycommunications@manchester.ac.uk providing relevant details, so we can investigate your claim.



DOI: 10.1002/cl2.1408

EVIDENCE AND GAP MAP

Ageing



In-person interventions to reduce social isolation and loneliness: An evidence and gap map

Vivian Welch¹ | Elizabeth Tanjong Ghogomu² | Sierra Dowling¹ | Victoria I. Barbeau¹ | Ali A. A. Al-Zubaidi³ | Ella Beveridge⁴ | Mostafa Bondok⁵ | Payaam Desai⁶ | Rebecca Doyle¹ | Jimmy Huang⁶ | Tarannum Hussain⁶ | Alyssa Jearvis⁶ | Fatima Jahel⁷ | Leen Madani⁶ | Wan Yuen Choo⁸ | Raudah M. Yunus⁹ | Tengku A. M. Tengku Mohd¹⁰ | Arpana Wadhwani⁴ | Abdulah Al Ameer⁴ | Rayan Ibrahim⁴ | Sarah Allam¹¹ | Niobe Haitas¹² | Sivan Bomze¹³ | Simone Dahrouge¹⁴ | Edward Garcia¹⁵ | Julianne Holt-Lunstad¹⁶ | Mathias Lasgaard¹⁷ | Michelle H. Lim^{18,19} | Kate Mulligan²⁰ | Douglas M. Salzwedel²¹ | Pamela Qualter²² | Paul C. Hébert²³ | Christopher Mikton²⁴

Correspondence

Vivian Welch, Bruyère Research Institute, Ottawa, Canada. Email: vwelch@campbellcollaboration.org

Funding information World Health Organization

Abstract

Background: Social isolation and loneliness can occur in all age groups, and they are linked to increased mortality and poorer health outcomes. There is a growing body of research indicating inconsistent findings on the effectiveness of interventions aiming to alleviate social isolation and loneliness. Hence the need to facilitate the discoverability of research on these interventions.

Objectives: To map available evidence on the effects of in-person interventions aimed at mitigating social isolation and/or loneliness across all age groups and settings.

Search Methods: The following databases were searched from inception up to 17 February 2022 with no language restrictions: Ovid MEDLINE, Embase, EBM Reviews—Cochrane Central Register of Controlled Trials, APA PsycInfo via Ovid, CINAHL via EBSCO, EBSCO (all databases except CINAHL), Global Index Medicus, ProQuest (all databases), ProQuest ERIC, Web of Science, Korean Citation Index, Russian Science Citation Index, and SciELO Citation Index via Clarivate, and Elsevier Scopus.

Selection Criteria: Titles, abstracts, and full texts of potentially eligible articles identified were screened independently by two reviewers for inclusion following the outlined eligibility criteria.

For affiliations refer to page 30.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

^{© 2024} The Authors. Campbell Systematic Reviews published by John Wiley & Sons Ltd on behalf of The Campbell Collaboration.

Data Collection and Analysis: We developed and pilot tested a data extraction code set in Eppi-Reviewer. Data was individually extracted and coded. We used the AMSTAR2 tool to assess the quality of reviews. However, the quality of the primary studies was not assessed.

Main Results: A total of 513 articles (421 primary studies and 92 systematic reviews) were included in this evidence and gap map which assessed the effectiveness of inperson interventions to reduce social isolation and loneliness. Most (68%) of the reviews were classified as critically low quality, while less than 5% were classified as high or moderate quality. Most reviews looked at interpersonal delivery and community-based delivery interventions, especially interventions for changing cognition led by a health professional and group activities, respectively. Loneliness, wellbeing, and depression/ anxiety were the most assessed outcomes. Most research was conducted in high-income countries, concentrated in the United States, United Kingdom, and Australia, with none from low-income countries. Major gaps were identified in societal level and communitybased delivery interventions that address policies and community structures, respectively. Less than 5% of included reviews assessed process indicators or implementation outcomes. Similar patterns of evidence and gaps were found in primary studies. All age groups were represented but more reviews and primary studies focused on older adults (≥60 years, 63%) compared to young people (≤24 years, 34%). Two thirds described how at-risk populations were identified and even fewer assessed differences in effect across equity factors for populations experiencing inequities.

Authors' Conclusions: There is growing evidence that social isolation and loneliness are public health concerns. This evidence and gap map shows the available evidence, at the time of the search, on the effectiveness of in-person interventions at reducing social isolation and loneliness across all ages and settings. Despite a large body of research, with much of it published in more recent years, it is unevenly distributed geographically and across types of interventions and outcomes. Most of the systematic reviews are of critically low quality, indicating the need for high quality reviews. This map can guide funders and researchers to consider the areas in which the evidence is lacking and to address these gaps as future research priorities.

1 | PLAIN LANGUAGE SUMMARY

1.1 | This evidence and gap map on in-person interventions to reduce social isolation and loneliness across all age groups found few studies of societal-level interventions, most research was from high income countries and the available reviews were of low quality

1.1.1 | The EGM in brief

This report discusses the effectiveness of in-person interventions to reduce social isolation and loneliness in people across all ages. It emphasizes that the available evidence is not consistent and varies in support of interpersonal and community-based delivery interventions and health-related outcomes (loneliness, wellbeing, depression/anxiety), with most evidence coming from high-income countries.

1.1.2 | What is this EGM about?

Social isolation and loneliness are becoming more significant public health issues associated with increased deaths and poorer health outcomes.

In-person interventions can help people who are at risk of feeling socially isolated and lonely by teaching them how to maintain their existing connections or form new ones. Policy makers are interested in knowing which interventions are effective in reducing social isolation and loneliness.

1.1.3 | What is the aim of this EGM?

To present available evidence from systematic reviews and primary studies about the effects of in-person interventions to reduce social isolation and/or loneliness across all ages and settings.

1.1.4 | What studies are included?

The EGM included 513 articles comprising 421 primary studies and 92 systematic reviews that evaluate the effectiveness of in-person interventions for reducing social isolation and/or loneliness. The studies had to report on the effect of in-person interventions and could be conducted in any setting.

1.1.5 | What are the main findings of this gap map?

The evidence in this EGM varies with more evidence for interpersonal and community-based delivery interventions than interventions delivered at societal levels. Most measured outcomes include health-related outcomes of loneliness, well-being, and depression/anxiety. Specifically, there are evidence clusters for community-based interventions of group activities and interpersonal delivery interventions for changing negative thinking and feelings about social relationships. Most evidence comes from high-income countries including the US, UK, and Australia, with none from low-income countries. Most systematic reviews included in this EGM are of critically low quality, with less than 5% of high or moderate quality.

Less than 5% of included reviews reported outcomes about how well the interventions were implemented. Only two primary studies and one review assessed interventions delivered at the societal level and only one review assessed interventions for age-friendly communities.

While all age groups were represented, more reviews and primary studies were interested in older adults (60 years or above, 63%) than young people (24 years or below, 35%). Very few described how they identified people who were at-risk and even fewer assessed whether the effects of the interventions were different for populations experiencing inequities.

1.1.6 | What do the findings of the map mean?

This EGM is a step toward identifying effective in-person interventions for reducing social isolation and loneliness for all age groups. Though a large body of evidence exists, with much of this research from more recent years, it shows variations, with a lack of high quality reviews. The map contributes to building an evidence base in the field by which funders and researchers can identify gaps in the evidence and then address these as future research priorities.

Collaboration

1.1.7 | How up-to-date is this EGM?

We searched for relevant studies up until 17 February 2022.

2 | BACKGROUND

2.1 | Introduction

2.1.1 | The problem, condition, or issue

Social isolation is the objective lack or paucity of social contact and infrequent interactions with others (Badcock et al., 2022; Donovan et al., 2020; Leigh-Hunt et al., 2017). Loneliness is a related concept defined as the subjective, negative feeling of inadequate meaningful connections resulting from an unmet need or discrepancy between desired and actual social relationships (Perlman et al., 1981; Prohaska et al., 2020). Loneliness has two components: an emotional component (unpleasant, negative feeling) and a social cognition component (perception of social disconnection from other people with a desire to be connected) (Badcock et al., 2022). Loneliness can also be a transient normal experience or chronic with negative physical and mental health consequences (Akhter-Khan et al., 2020; Qualter et al., 2015).

The two concepts, social isolation, and loneliness, are distinct; social isolation is objective and associated with social contact while loneliness is subjective and related to social connectedness (O'Rourke et al., 2018). One may occur without the other, although they are related and may also co-occur. People may have a large social network and feel lonely, while some with a small network may not.

Social isolation and loneliness can occur across all age groups and are associated with serious health consequences including anxiety and depression, cardiovascular disease, and premature mortality (Cené et al., 2022; Leigh-Hunt et al., 2017). One German cohort study of nearly 5000 middle-aged and older adults found that the risk of mortality is exacerbated when social isolation and loneliness cooccur (Beller et al., 2018). Poor relationships are associated with 32% increased risk of stroke, 29% increased risk of coronary heart disease (Cené et al., 2022; Valtorta et al., 2016), and 26% increased risk of allcause mortality (Donovan et al., 2020; Holt-Lunstad et al., 2015). Incident depression and dementia have a bi-directional relationship with both social isolation and loneliness (Cené et al., 2022; Donovan et al., 2020) although several studies have reported that dementia is associated with loneliness than social isolation (Cené et al., 2022; Holwerda et al., 2014; Rafnsson et al., 2020). Both social isolation and loneliness are associated with negative health-related behaviors such as smoking and physical inactivity (Cené et al., 2022; Menec et al., 2020). The negative health impacts of social isolation and loneliness have been shown to increase health and social care service use (Cotterell et al., 2018; Windle et al., 2012). These negative impacts occur when contextual and risk factors affecting social relationships persist and individuals do not use appropriate coping

strategies to address them (Akhter-Khan et al., 2022; Elder et al., 2012).

Since the onset of the COVID-19 pandemic, movement restriction policies have made social isolation and loneliness prominent global issues and a public health priority (Galvez-Hernandez et al., 2022; Holt-Lunstad 2022; WHO, 2021). The prevalence of severe loneliness increased by 15% and social isolation by 13% in adults 18 years or older across 101 countries during the pandemic (O'Sullivan et al., 2021). Small increases in the prevalence of loneliness were also observed in a recent synthesis of longitudinal studies during the COVID-19 pandemic (Ernst et al., 2022). The prevalence is hard to measure across the lifespan because of the lack of standardized measurement instruments and definitions, and the use of different cut-off points and age categories (Holt-Lunstad, 2022; Prohaska et al., 2020). A recent systematic review and meta-analysis on the prevalence of loneliness pre-COVID-19 pandemic across 113 countries (Surkalim et al., 2022) showed varying rates for adolescents (9.2% to 14.4%), young adults (1.8% to 9.4%), middle-aged adults (2.4% to 12%), and older adults (4.2% to 24.2%) depending on the country. A prevalence study in a German population-based adult cohort showed that social isolation increases with age from 5.4% (95% confidence interval [CI]: 4.7 to 6.0) in the youngest age group (18 to 39 years) to 21.7% (95% CI: 19.5 to 24.0) in the oldest age group (70 to 79 years) (Röhr et al., 2021). The global prevalence of social isolation in community-dwelling older adults was found to be 25% (95% CI: 21 to 30) (Teo et al., 2022). Studies have found a u-shaped association between age and loneliness with the highest levels of loneliness in younger and older people (Lasgaard et al., 2016; Yang et al., 2011), but few studies have focused on vounger people. Although social isolation and loneliness are global issues, most of the studies were conducted in high income countries, especially in Europe, with very few in low-middle-income countries (Fakoya et al., 2020; Surkalim et al., 2022).

Both social isolation and loneliness are linked to less social support and can be triggered by situational factors such as adversity, significant life changes or transitions, such as moving away from home, starting a new job, becoming a parent, illness, and the death of a spouse or parent (Badcock et al., 2022; Elder et al., 2012; Lim et al., 2020; Qualter et al., 2022). They are associated with risk factors including individual factors (e.g., personality, maladaptive cognition, poor health, disability or mobility impairment, cognitive impairment), interpersonal or social factors (e.g., peer victimization or discrimination, poor relationship quality, quantity of friends or social contacts, living alone), socio-environmental factors (e.g., neighborhood deprivation, inaccessible location of residence, housing, cultural prejudice), and demographic factors (e.g., age, gender, educational level, low socio-economic status, unemployment) (Badcock et al., 2022; Elder et al., 2012; Lim et al., 2020; O'Sullivan et al., 2021; Qualter et al., 2022).

Many systematic reviews have evaluated the effectiveness of interventions to reduce social isolation or loneliness with conflicting findings demonstrating a need for better quality research (Masi et al., 2011; Victor et al., 2018; Williams et al., 2021). A number have

focused on older adults, but social isolation and loneliness affect people across the life span, including young people (Qualter et al., 2015; Surkalim et al., 2022), with interventions designed specifically for them (Eccles et al., 2021). Most of the reviews have focused on people living in the community or long-term care settings (Fakoya et al., 2020; Grenade et al., 2008). There is limited research addressing social isolation and/or loneliness for patients in clinical settings (NASEM, 2020). Studies that consider hospitalized patients focus on screening and detection of loneliness and social isolation, the impact of social isolation and loneliness on health service use and which interventions may be used; hardly any focus on assessing the effectiveness of interventions to reduce social isolation and loneliness (Grenade et al., 2008; NASEM, 2020; Proffitt et al., 1993; Razai et al., 2020; Zamir et al., 2018).

The impact of interventions has been found to differ depending on population characteristics such as coping skills, needs, degree of loneliness, and contextual factors like age, socioeconomic status, health condition, and place of residence (Fakoya et al., 2020). Therefore, there is no one-size-fits-all approach, and it is important to tailor appropriate interventions to individuals' needs and contexts (Akhter-Khan et al., 2020; Fakoya et al., 2020; Mann et al., 2017).

There are health equity issues related to social isolation and/or loneliness such as the gap in evidence from low-middle income countries (Surkalim et al., 2022), limited access to interventions caused by disabilities and lack of transportation, or limited programs in rural areas compared to urban areas (Dassieu et al., 2021; NASEM, 2020; Qualter et al., 2022). Social isolation and/or loneliness related to structural inequities (e.g., intersectional discrimination across race, gender, socioeconomic status; age-based discrimination and ethnic minorities), have a negative impact on health outcomes (Dassieu et al., 2021).

This current evidence and gap map identifies areas where evidence is available, as well as any gaps in research related to inperson interventions for social isolation and loneliness across any age.

2.1.2 | The intervention

Different types of interventions for reducing social isolation (Dickens et al., 2011; Findlay, 2003), loneliness (Cohen-Mansfield et al., 2015; Eccles et al., 2021; Hagan et al., 2014; Jarvis et al., 2020; Mann et al., 2017; Masi et al., 2011; Veronese et al., 2021), or both social isolation and loneliness (Cattan et al., 2005; Gardiner et al., 2018; Poscia et al., 2018) have been described and assessed in several systematic and scoping reviews. However, there is a lack of a standardized framework for describing these interventions (Fakoya et al., 2020; Prohaska et al., 2020). Interventions for reducing social isolation and loneliness are often complex with multiple and interacting components, working through different potential mechanisms of action (Fakoya et al., 2020; Gardiner et al., 2020;

Several approaches have been used to categorize interventions in some reviews. The interventions have been categorized by the format or delivery mode or type as one-on-one or group-based (Cohen-Mansfield et al., 2015; Dickens et al., 2011; Fakoya et al., 2020; Findlay, 2003; Hagan et al., 2014; Masi et al., 2011; Poscia et al., 2018), or technology or non-technology (in-person) based (Eccles et al., 2021; Masi et al., 2011). They have also been categorized by the type, or strategy, being classified as interventions for social skills training, enhancing social support, enhancing social interaction or social cognition training (Masi et al., 2011). Other terms have been used as a rationale for categorization, such as the focus, nature or goal of the intervention (Cohen-Mansfield et al., 2015; Fakoya et al., 2020; Masi et al., 2011); the purpose, intended outcomes, and mechanisms by which interventions target social isolation and loneliness (Fakoya et al., 2020; Gardiner et al., 2018).

The scoping review by Mann et al. classifies interventions as direct (i.e., targeting loneliness and related concepts) or indirect (targeting wellbeing which impacts loneliness), and also articulates various levels of engagement for those delivering the interventions following the socio-ecological model: individual level, relationship and community level, and societal level (Mann et al., 2017).

Another scoping review (O'Rourke et al., 2018) classified interventions for reducing loneliness by their components into nine types: one-to-one personal contact, activity group, animal contact, skills course interventions, or varied/non-specific interventions, reminiscence, support group, model of care and public broadcast.

This evidence and gap map focuses on in-person interventions that are non-technology based and delivered face-to-face since there is a gap map on digital interventions for older adults (Welch et al., 2022b).

2.1.3 | Why it is important to develop the EGM

The existing body of evidence for interventions to mitigate social isolation and/or loneliness is rapidly growing and characterized by small, low-quality trials, with inconsistent terminology and conclusions on their effectiveness (Eccles et al., 2021; Fakoya et al., 2020; Prohaska et al., 2020; Veronese et al., 2021) as well as equity-related issues. With the rapid growth of evidence in this sector, this evidence and gap map will demonstrate areas where evidence is available and areas where there are gaps that researchers, decision and policy-makers could use to help select interventions and prioritize future research to fill gaps in research and to understand which interventions might work in which conditions. It will also improve the discoverability of evidence on different types of interventions and enhance their use for informed decision-making by stakeholders including health and social care providers, policymakers, citizens, caregivers, and patients.

2.1.4 | Existing evidence and gap maps and/or relevant systematic reviews

There is rapidly expanding research on alleviating social isolation and/or loneliness since the COVID-19 pandemic. Several relevant systematic



5 of 40

reviews have been conducted and included in three umbrella reviews (Boulton et al., 2021; Jarvis et al., 2020; Veronese et al., 2021). Boulton et al. found mixed evidence of effectiveness on loneliness for remote befriending, social support, and low intensity psychosocial interventions. Jarvis assessed various interventions addressing loneliness in older adults and found limited effect, with the greatest effect in a social cognition intervention. Veronese et al. reported low or very low-quality evidence of three interventions (meditation/mindfulness, social cognitive training, and social support) that reduced loneliness. A scoping review of reviews showed the lack of studies conducted in low-middle income countries (Fakoya et al., 2020).

One evidence and gap map has been conducted on remotely delivered digital interventions including befriending, social support, and low-intensity psychosocial interventions for social isolation and loneliness in older adults (Boulton et al., 2021). It showed mostly lowquality reviews and few studies on older people who are not caregivers or who do not have a particular chronic illness. Our group is currently working on another gap map on digital interventions for older adults with a broader scope of interventions and including caregivers (Welch et al., 2022; Welch et al., 2022b), but there is currently no mapping of evidence for in-person interventions to reduce social isolation and loneliness across all ages.

3 | OBJECTIVES

This EGM aims to map available evidence on the effects of in-person interventions to reduce social isolation and/or loneliness across all age groups in all settings.

Specific objectives are as follows:

- To identify existing evidence from primary studies and systematic reviews on the effects of in-person interventions that are nontechnology based and delivered face-to-face to reduce social isolation and/or loneliness across all age groups.
- To identify research evidence gaps for new high-quality primary studies and systematic reviews.
- 3. To highlight evidence of health equity considerations from included primary studies and systematic reviews.

4 | METHODS

We followed the Campbell Collaboration guidance for producing an evidence and gap map (White et al., 2020) described in our protocol (Welch et al., 2023).

4.1 | Evidence and gap map: Definition and purpose

Evidence and gap maps are a systematic evidence synthesis product with a visual presentation of existing evidence relevant to a specific research question (Snilstveit et al., 2013; White et al., 2020). They display areas where evidence is available, areas where there are gaps in evidence, and the quality of existing evidence.

WILEY-Campbell Collaboration

The evidence and gap map is typically a two-dimensional matrix with interventions as row headings and outcomes as column headings (Snilstveit, 2016; White et al., 2020). The studies with evidence on the corresponding intervention and outcome are shown within each cell of the matrix. This map will identify areas of evidence and any gaps in research related to using in-person interventions for social isolation and/or loneliness across all ages.

4.2 | Framework development and scope

We developed an intervention-outcome framework by adapting our conceptual framework from the digital interventions EGM (Welch et al., 2022). We expanded the non-digital intervention and outcome categories to attain evidence-based, clear and distinct categories that are practical and useful to a broad audience by using several existing frameworks, reports, and reviews.

In consultation with stakeholders, we identified and reviewed other frameworks including the framework described by Masi (Masi et al., 2011), the framework for the Campaign to End Loneliness by Jopling (Jopling, 2020), the framework by Mann (Mann et al., 2017), the socio-ecological framework adapted by the World Health Organization for strategies to reduce social isolation and loneliness (WHO, 2021), the framework for evidence-based interventions for youth reporting loneliness (Qualter et al., 2022), and the social relationship expectations framework (Akhter-Khan et al., 2022). We also considered the National Academies of Sciences, Engineering, and Medicine Consensus study report on social isolation and loneliness in older adults (NASEM, 2020), the American Association of Retired Persons (AARP) Foundation report on social isolation (Elder et al., 2012), the taxonomy to evaluate social isolation and loneliness interventions developed by the Foundation for Social Connection, and three reviews on loneliness (Holt-Lunstad, 2022; Lim et al., 2020; Ogrin et al., 2021).

Six of the existing frameworks and reviews have adopted the socio-ecological framework when considering interventions to reduce social isolation and/or loneliness (Holt-Lunstad, 2022; Lim et al., 2020; Mann et al., 2017; Ogrin et al., 2021; Qualter et al., 2022; WHO, 2021). The socio-ecological framework has been used to explore the complex dimensions of other public health issues like violence and abuse (CDC, 2015; WHO, 2002) and health promotion (Wold et al., 2018). The model examines the relationship between risk and protective factors at different levels of influence including individual, relationship, community, and societal levels. The individual level focuses on personal characteristics that increase risks. The relationship level focuses on risk factors involving close social relationships with family and friends. The community level explores risk factors from wider social relationships in community settings such as schools, workplaces, and neighborhoods. The societal level of influence considers broad societal risk factors such as policies and cultural norms.

Social isolation and loneliness are associated with the lack of meaningful social connections which can occur at any of these four levels (Holt-Lunstad, 2018, Holt-Lunstad, 2022; Lim et al., 2020; Ogrin et al., 2021). Risk factors for social isolation and loneliness can be co-occurrent, inter-related, and can operate at multiple levels (Holt-Lunstad, 2022; Lim et al., 2020; Qualter et al., 2022). Interventions may target risk factors at multiple levels of the socio-ecological model by creating and maintaining meaningful social connections or a combination of other mechanisms, such as changing negative social cognition (i.e., changing negative thinking and feelings about social relationships) or providing support to enhance social interactions.

We considered non-technology-based interventions delivered inperson to alleviate social isolation and/or loneliness across all age groups in all settings. None of the existing frameworks or taxonomies provide mutually exclusive categories and subcategories for classifying the interventions for this evidence and gap map. They all demonstrate the complexity, diversity, and interdependencies of contextual or risk factors, and mechanisms that shape social relationships.

We therefore focused on delivery and classified interventions into five main categories based on who is providing the intervention and where it is provided: self-delivery, interpersonal delivery, communitybased delivery, societal-level delivery, and multi-component or complex interventions. These intervention categories are further described in the conceptual framework.

Outcomes were based on the level of impact of interventions. The impact of social isolation and loneliness interventions depend on how well they were implemented, therefore, we considered both process indicators or implementation outcomes and other outcomes including health and psychosocial outcomes, indicators of social connections as well as cost and cost-effectiveness outcomes (Jopling, 2020; Windle et al., 2012). See Glossary of terms in Supporting Information Appendix 1.

4.3 | Stakeholder engagement

We established an Advisory Board of key stakeholders to contribute toward defining the scope and developing the framework for the map as well as interpreting the findings. They include academics, advocates, policy and decision-makers, from relevant organizations (e.g., WHO, Canadian Red Cross, Global Initiative on Loneliness and Connection, US Foundation for Social Connection, and Ending Loneliness Together) who are involved in research and working to address social isolation and loneliness.

The Advisory Group met virtually on December 13 and 16, 2021 to discuss the scope of the evidence and gap map and existing frameworks that could be considered in developing the interventionoutcome framework for this evidence gap map. They met again in June 2022 to provide feedback on the framework. They were consulted by email to provide feedback on the revised framework and on the preliminary findings and draft map.

4.4 | Conceptual framework

The conceptual framework (Figure 1) considers possible pathways for interventions to bring about expected changes and outcomes based on the understanding of the population risk factors and needs that may trigger social isolation or loneliness. It is based on theoretical underpinnings with the following key components:

- population contexts, risk factors, and needs that may trigger social isolation or loneliness.
- types of interventions required to address social isolation or loneliness,
- the mechanisms of change by which the interventions address social isolation or loneliness, and
- process indicators (e.g., acceptability) and outcomes (e.g., loneliness).

4.4.1 | Population targeted by interventions

Social isolation and loneliness are complex public health issues and their occurrence across the lifespan is influenced by individual contextual and risk factors, needs, expectations, and coping skills which are all inter-related and influence relationship ties (Akhter-Khan et al., 2022; Elder et al., 2012; Gardiner et al., 2018; O'Rourke et al., 2018; Qualter et al., 2022). Contextual and risk factors such as structural changes that may cause displacement (e.g., moving schools



or wars), living situations (e.g., living alone or in a care facility such as orphanage, long-term care home), resources (available activities or social supports) affect people's motives, expectations, coping skills, and social relationships. Coping skills and social supports may be a protective factor if they allow people to promote their wellbeing or resilience. On the other hand, inadequate coping skills and social supports may be a risk factor for social isolation or loneliness. Based on a public health approach, interventions may target

anyone regardless of risk (universal), or target subpopulations at high risk (selective) or socially isolated or lonely people (indicated) (Springer et al., 2007). Categorizing target populations into these three orders gives a clearer picture and understanding of whom to prioritize and how to allocate resources more efficiently.

4.4.2 | Risk factors

Social isolation or loneliness may be triggered in both young and old across the life span by situational factors such as significant life events or transitions (e.g., adversity, moving away from home, retirement, death of a spouse, friend, or relative) and may be associated with risk factors including physical and mental health factors (e.g., poor health, maladaptive cognition or cognitive decline, disability or impaired mobility, personality), interpersonal or social factors (e.g., living alone, peer victimization, social contacts, relationship quality), socio-cultural or environmental factors (e.g., neighborhood deprivation, inaccessible location of residence, cultural

Population - across the life span	Interventions Self delivery	Mechanisms	Outcomes
Population targeted by interventions	- Changing cognition (e.g., gratitude writing exercise) - Social skills training and psychoeducation	Support for enhancing	Health and psychosocial
Universal (regardless of risk) Selective (high risk) Indicated (socially isolated or	Interpersonal delivery - Changing cognition (e.g., reminiscence therapy) - Social skills training and psychoeducation	social connections	outcomes e.g., loneliness, social isolation, social connectedness, social
Risk factors	 Healthcare support (e.g., referral to social services) Social support (e.g., befriending programs) 	Maintain connections	support, quality of life/well- being, self-efficacy, anxiety/depression, adverse
Situational factors (life events and transitions e.g., adversity, moving away from home), physical and mental health factors (disability, maladaptive cognition), interpersonal or social factors (e.g., living alone, peer victimization), socio-	Community-based delivery - Group activities (e.g., gardening, arts, exercise) - Support groups (e.g., bereavement, caregivers) - Neighborhood approaches (e.g., networks, transportation, intergenerational approaches) - Age-friendly communities (e.g., friendly schools) - Volunteering	New connections	effects, barriers, health services use (emergency room visits, hospitalization, premature institutionalization)
		Change in negative	Indicators of social connections
cultural/environmental factors (e.g., neighborhood deprivation), demographic factors (e.g., age, gender)	Societal level delivery - Public health policy that promotes social connection and addresses social isolation/loneliness	social cognition	e.g., social support, social engagement, social cohesion, social capital
Needs assessment Practical/basic, Social,	 Public education and awareness on developing and maintaining meaningful social connections/relations Policies on the design of neighbourhoods 	e.g., acceptance, adherence, feasibility,	Cost & cost- effectiveness outcomes
Emotional, Clinical/health, Financial, Education, Communication, Caregiver	Multi-component/complex e.g., reminiscence therapy and group activities	increased awareness of community services, barriers	

WILEY-Campbell Collaboration

individualism, social discrimination, and stigma) and demographic factors (e.g., age, gender, socio-economic status) (Dahlberg et al., 2022; Elder et al., 2012; Lim et al., 2020; NASEM, 2020; Qualter et al., 2022).

4.4.3 | Needs assessment

Socially isolated and lonely individuals across the lifespan may have unmet needs that can be due to low social support or disruption in social interactions with people at different levels—individual, group, community, or societal (Elder et al., 2012; Holt-Lunstad, 2022; NASEM, 2020; WHO, 2021). These needs may include basic needs (housing, nutrition and food security, and healthcare), social and emotional needs (social connections and companionship), financial, education (skills development and learning), communication, caregiver needs, home modifications and maintenance, domestic assistance, mobility, personal care, respite care and civic engagement (meaningfulness and status, the need for having a purpose in later life or being able to contribute usefully to society) (Abdi et al., 2019; Bedney et al., 2010; Bee et al., 2014; Eccles et al., 2021).

Social support is a major component of social connection and may be provided to meet different needs. Social support can take the forms of instrumental/tangible, informational, emotional, or belonging support (Elder et al., 2012; Holt-Lunstad, 2022; NASEM, 2020). It can be perceived, or actual support provided through social connections with other people and through different sectors including health, transportation, housing, work, nutrition, environment, education, leisure: arts and entertainment (Holt-Lunstad, 2022).

Loneliness is also associated with social relationship expectations that are influenced by personal, social, cultural, and historical contexts and include the availability of social contacts (proximity), feeling cared for and relying on others (support), feeling close, understood, and listened to (intimacy), sharing interests and enjoyable experiences (fun), having opportunities to contribute meaningfully (generativity) and feeling valued and actively included (respect) (Akhter-Khan et al., 2022). A discrepancy between expected and actual social relationships will result in loneliness.

It is important to assess and understand individuals' specific contexts, risks, expectations, and needs to tailor appropriate interventions to reduce social isolation or loneliness (Akhter-Khan et al., 2022; Eccles et al., 2021; Fakoya et al., 2020; Jopling, 2020; Lim et al., 2020; Mann et al., 2017; NASEM, 2020).

4.4.4 | Interventions

Interventions to reduce social isolation and/or loneliness are more effective when targeted to the individual's specific experience and context (such as triggers, risk factors, and accessibility to resources) (Eccles et al., 2021; Fakoya et al., 2020; Prohaska et al., 2020; Veronese et al., 2021). Interventions aimed at reducing social isolation may have different components than those aimed at reducing

loneliness (O'Rourke et al., 2018). The interventions can be delivered through various modes to bring about changes at different levels individual, relationship, community, and societal levels (Holt-Lunstad, 2022; Lim et al., 2020; Mann et al., 2017; Ogrin et al., 2021; WHO, 2021). In addition, one component of an intervention may target multiple factors (O'Rourke et al., 2018). It is therefore challenging to categorize interventions by the risk factors they target following the socio-ecological framework or by their mechanisms. In order to have mutually exclusive categories and subcategories on the evidence map, we classified interventions as follows, based on who is providing the intervention and where the intervention is provided.

Self-delivery: These are self-guided interventions that focus on addressing personal characteristics (biological and sociodemographic factors, e.g., socio-economic status) that may trigger social isolation or loneliness, through strategies that change the attitudes, beliefs, and behaviors of individuals. The purpose of these interventions is to affect changes in an individual that would result in reductions in isolation and loneliness. They can be facilitated with training, guidance, or available resources from the healthcare professionals. Examples include self-guided social cognitive interventions like cognitive behavioral therapy, mindfulness, or reminiscence therapy.

Interpersonal delivery: These are interventions that focus on building close meaningful personal relationships with family and others in the community, for example, friends, colleagues, neighbors, volunteers. They can be delivered by healthcare or social care professionals, volunteers, or other people in the individual's social network. The purpose/aim of these interventions is to affect changes in an individual or a specific relationship or network of relationships that would result in reductions in isolation and loneliness. They are accomplished through mechanisms including changing cognition, social skills training and psychoeducation, healthcare support, and social support. Examples include cognitive behavioral therapy, family therapy, social prescribing or friendship enrichment program.

Community-based delivery: These interventions are delivered by healthcare or social care professionals, community workers/volunteers, or other people within the same community setting. They can be delivered through neighborhood organizations, community-based healthcare, and social services or facilities. They focus on addressing risk factors in social settings and increasing opportunities for social interactions with others such as connecting to community group activities or peer support groups; neighborhood approaches like networks, transportation, meals on wheels; intergenerational approaches. They may also provide a supportive environment or encourage participation by improving access to amenities within the community e.g., built environment, age-friendly communities, and volunteering. These interventions can affect changes in an individual, a specific relationship or network of relationships, or changes in the community that would result in reductions in isolation and loneliness.

Societal level delivery: Focus on policies and laws that address societal risk factors like discrimination and stigma, socio-economic inequality or may seek to change social norms that prevent social connection such as policies addressing housing, employment, transportation and the environment. These interventions affect change(s) in broader society, resulting in reductions in isolation and loneliness. Examples include public awareness campaigns, coalition and partnership initiatives, or family-friendly policies.

Multi-component/complex: These are combinations of multiple components within the interventions involving the same/different types of delivery modes in the same study, for example, reminiscence therapy and group activities.

4.4.5 | Mechanisms

Different pathways or mechanisms related to contextual or risk factors, motives, expectations, and coping skills have been identified through which interventions may reduce social isolation or loneliness. The interventions may improve and maintain existing relationships or enable people to create new connections (Akhter-Khan et al., 2022; Jopling, 2020; Mann et al., 2017) by addressing contextual or risk factors (Akhter-Khan et al., 2022; Lim et al., 2020; Ogrin et al., 2021). Some interventions aim to change one's outlook or negative social cognition (Akhter-Khan et al., 2022; Jopling, 2020; Mann et al., 2017; Masi et al., 2011; Ogrin et al., 2021) while others provide support to enhance social interactions (Mann et al., 2017; Masi et al., 2011; Ogrin et al., 2021). Some interventions involve building skills, purposeful activity, or implementing a philosophy of care within a facility (Akhter-Khan et al., 2022). Some interventions are complex and may address social isolation or loneliness through multiple or poorly specified mechanisms (Akhter-Khan et al., 2022; Holt-Lunstad, 2018; Lim et al., 2020).

4.4.6 | Process indicators and outcomes

The potential of interventions to reduce social isolation or loneliness have been assessed through their acceptability, adherence, and feasibility. These process indicators determine progress toward outcomes such as health and psychosocial outcomes (e.g., loneliness, social isolation, social connectedness), indicators of social connections (e.g., social support, social engagement, social cohesion), as well as cost and cost-effectiveness outcomes. See Glossary of terms (Supporting Information Appendix 1).

We used this conceptual framework to define and code the intervention and outcome categories and subcategories for the twodimensional matrix in the evidence and gap map.

4.5 | Dimensions

4.5.1 | Types of study design

We included on-going (protocols, trial registrations) and completed systematic reviews and primary studies with any study design that has a control group: randomized controlled trials, non-randomized studies including control before-after, and statistical matching quasiexperimental studies.



The inclusion of systematic reviews was based on the population, intervention, comparison, outcome (PICO) framework and if they met at least four of the five criteria of a systematic review as defined by Moher et al. (2015). That is, they describe adequate search methods used to identify studies, eligibility criteria for selection of studies, methods of critical appraisal of included studies, sufficient details or characteristics of included studies, and synthesis or analysis of findings of the included studies.

Eligible quasi-experimental designs included quasi-randomized studies, regression discontinuity designs, natural experiments, non-equivalent comparison group designs and interrupted series designs with at least three data points before and after a discrete intervention (Waddington et al., 2014).

We excluded any study designs with no control group such as longitudinal cohort studies and cross-sectional studies, and those studies with interrupted time series designs with less than six data points.

We included mixed methods studies, but exclusive qualitative research was excluded.

We included studies irrespective of their publication status (e.g., protocols, working papers).

4.5.2 | Types of intervention/problem

We included any intervention which aims to reduce social isolation and/or loneliness that is delivered in-person regardless of the intensity, duration, and frequency of administration. We excluded digital or technology-based interventions.

Included interventions may be one-on-one or group based and were categorized based on our conceptual framework as selfdelivery, interpersonal delivery, community-based delivery, societal-level delivery as well as multi-component or complex interventions. See Table 1 for subcategories and examples.

Comparison interventions included no interventions, other interventions, or usual care.

If reviews included a subset of interventions that is not eligible, we only coded studies with the eligible interventions.

4.5.3 | Types of population

We considered any age group, people with or at risk of social isolation or loneliness, or the general population, whether based on case finding or screening for vulnerability or not.

- <10 years (children)
- 10-24 years (adolescents/youth)
- 25-44 years (young adults)

Age groups include:

- 44-60 years (middle-aged)
- 60–75 years (youngest-old)

18911803, 2024, 2, Downloaded from https://anlinelibrary.wiley.com/doi/10.1002/cl2.1408 by The University Of Manchester, Wiley Online Library on [02:07/2024]. See the Terms and Conditions (https://anlinelibrary.wiley.com/doi/10.1002/cl2.1408 by The University Of Manchester, Wiley Online Library on [02:07/2024].

and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

10 of 40 WILEY- Collaboration

TABLE 1Types of interventions.

Intervention categories	Subcategories	E>	amples
Self-delivery	Self-guided changing cognition	-	Self-guided mindfulness therapy Self-guided reminiscence therapy
	Self-guided social skills training and psychoeducation	-	Solitary social skills training Psychoeducation e.g., gratitude
Interpersonal delivery	Changing cognition led by a health professional		Cognitive behavioral therapy Mindfulness therapy Reminiscence therapy
	Social skills training and psychoeducation led by a health professional	- - -	Friendship enrichment program Family therapy Psychosocial school intervention
	Healthcare support	-	Hearing aids Social prescribing (Primary care referral to support services)
	Social support	-	Community navigators
Community-based delivery	Group activities	-	Activities aimed at bringing people together through shared interests as well as facilitating social connection, e.g., education or health promotional activities (gardening, exercise, or fitness program)
	Support groups	-	Group-based interventions for people with common conditions or risk factors for social isolation or loneliness, e.g., diabetes, bereavement, caregivers
	Neighborhood approaches		Community networks Intergenerational approaches Meals on wheels Lunch clubs Spiritual-based programs Built environment (changes to transportation infrastructure, housing and landscape design improvements, parks)
	Age-friendly communities	- -	Dementia friendly communities Friendly schools World Health Organization age-friendly communities
	Volunteering	-	Volunteering
Societal level delivery	Public health, healthcare, and social policies that promote social connection, address loneliness and social isolation, facilitate social cohesion and inclusion	-	Frome Model of Enhanced Primary Care
	Public education and awareness of how to develop and maintain meaningful social connection and relationships	-	Campaign to end loneliness
	Policies on the urban design of neighborhoods and social infrastructure of communities		
	Policies on workplaces and how to initiate, maintain, and develop meaningful social connection with co-workers and with the organization	-	Internal employer policies and procedures that foster employee connection (group spaces, peer activities, mentoring programs), and government incentives to foster organizational change (e.g., tax credits to induce behavior change).
	Funding relevant research on implementing programs and policies and facilitating the rapid translation from evidence to practice and policy	-	Funding relevant research on implementing programs and policies and to facilitate the rapid translation from evidence to practice and policy
Multi-component or complex	-	-	Social skills training and mindfulness group counseling

- 75-85 years (middle-old)
- >85 years (oldest-old)

4.5.4 | Types of outcome measures

We considered the following types of outcomes:

- health and psychosocial outcomes,
- indicators of social connections,
- · cost and cost-effectiveness outcomes, and
- · process indicators (or implementation outcomes).

We considered adverse effects of interventions such as psychological distress, safety, and others as described by the studies. Different measuring tools have been used for loneliness, social isolation, and related outcomes. See Table 2 for outcome categories and measurements.

We did not use outcomes as eligibility criteria; however, eligible studies and reviews must have assessed interventions with a primary objective to reduce social isolation and/or loneliness, and could have reported outcomes other than social isolation and loneliness. Studies and reviews were not included if the primary aim of the interventions assessed was not to reduce social isolation and/or loneliness. Those that assessed the effects of interventions on social isolation and/or loneliness as a primary outcome or considered other indicators of social connections, such as social support, social engagement, social cohesion, and social capital were also included.

Studies and reviews assessing the effect of interventions on indicators of psychological wellbeing such as quality of life, anxiety or depression, with a focus on mental health rather than social isolation or loneliness were excluded.

4.5.5 | Other eligibility criteria

Types of location/situation

We included all countries. We also classified the countries by the World Health Organization regions (African Region, Region of the Americas, South-East Asian Region, European Region, Eastern Mediterranean Region, Western Pacific Region) (WHO, 2019) and the World Bank classification by incomes: low income economies, lower-middle income economies, upper-middle income economies, and high income economies (World Bank, 2022).

We did not exclude primary studies and systematic reviews that do not report the countries.

Types of settings

We included all settings, for example, residential or personal home, nursing home or long-term care, assisted living facilities, orphanages, schools, workplaces, community centers, and medical facilities.

Collaboration

4.6 | Search methods and sources

An Information Specialist (DS) designed the search strategy which was peer-reviewed through PRESS (Peer Review of Electronic Search Strategies) (McGowan et al., 2016). See Supporting Information Appendix 2 for the full search strategies. We searched the following databases from inception to February 17, 2022 with no date or language restrictions: Ovid MEDLINE, Embase, EBM Reviews-Cochrane Central Register of Controlled Trials, Global Index Medicus. We searched CINAHL via EBSCOhost, and the following databases as a combined search via EBSCO (Academic Search Complete, Global Health, and AgeLine, America: History & Life; American Antiguarian Society (AAS) Historical Periodicals Collection: Series 1: American Antiguarian Society (AAS) Historical Periodicals Collection: Series 2; American Antiguarian Society (AAS) Historical Periodicals Collection: Series 3; American Antiquarian Society (AAS) Historical Periodicals Collection: Series 4; American Antiguarian Society (AAS) Historical Periodicals Collection: Series 5: Art Full Text (H.W. Wilson): Art Index Retrospective (H.W. Wilson); Atla Religion Database with AtlaSerials; Bibliography of Indigenous Peoples in North America; Book Review Digest Retrospective: 1903-1982 (H.W. Wilson); Business Periodicals Index Retrospective: 1913-1982 (H.W. Wilson); Business Source Complete; Communication Source; Criminal Justice Abstracts; eBook Collection (EBSCOhost): EconLit: Education Source: Ergonomics Abstracts: Essay and General Literature Retrospective (H.W. Wilson); Film & Television Literature Index with Full Text; Financial Times: FSTA - Food Science and Technology Abstracts: GreenFILE: Historical Abstracts; Humanities & Social Sciences Index Retrospective: 1907-1984 (H.W. Wilson); Index to Legal Periodicals & Books Full Text (H.W. Wilson): Index to Legal Periodicals Retrospective: 1908-1981 (H.W. Wilson); Index to Printed Music; International Bibliography of Theatre & Dance with Full Text; International Political Science Abstracts; Library Literature & Information Science Full Text (H.W. Wilson); Library Literature & Information Science Retrospective: 1905-1983 (H.W. Wilson); Library, Information Science & Technology Abstracts; MLA Directory of Periodicals; MLA International Bibliography; New Testament Abstracts; Old Testament Abstracts; Philosopher's Index; Readers' Guide Retrospective:: 1890-1982 (H.W. Wilson); Regional Business News; RILM Abstracts of Music Literature; Short Story Index Retrospective: 1915-1983 (H.W. Wilson);SPORTDiscus with Full Text).

We searched the ProQuest platform which includes 53 databases, the most directly relevant among them being: ASSIA, APA PsycArticles, PsycBooks, PsycInfo, Canadian Research Index, Eric, IBSS, Nursing and Allied Health, Publicly Available Content, Dissertations & Theses Global), the core collection of Web of Science including Conference Proceedings Citation Index, Korean Citation Index, Russian Science Citation Index, and SciELO Citation Index via Clarivate, and Elsevier Scopus.

In addition, we searched non-English databases including CNKI Chinese, ThaiJo (Thailand), CiNii-J-Stage (Japanese), CQVIP (China), Korean Social Science Data Center, MyCite (Malaysia). We translated terms for social isolation and loneliness and screened for relevant systematic reviews and controlled studies. -WILEY-Campbell Collaboration

TABLE 2Outcome categories.

Outcomes	Acceptable measurements
Health and psychosocial outcomes	
Loneliness	UCLA loneliness scale, de Jong-Gierveld loneliness scale, other scales, e.g., Social and Emotional Loneliness Scale, Hughes loneliness scale
Social isolation	Lubben's Social Network Scale, Social Network Index, PROMIS social isolation 6-I scale
Social connectedness/interactions/networks or life satisfaction	Lee and Robin's Social Connectedness Scale; Number of contacts; Frequency of social interactions; Satisfaction with interaction; Index of support satisfaction; Support network satisfaction; Companionship scale satisfaction
Well-being/Quality of life	MOS SF-36 Health Survey; Work and Social Adjustment Scale (WSAS); WHOQOL
Anxiety/depression	Beck Depression Inventory (BDI); Depression Adjective Check List (DACL) Form E; Geriatric depression scale; The Centre for Epidemiological Studies Depression Scale (CES-D)
Self-efficacy or self-esteem	General Self-Efficacy Scale, Rosenberg Self-Esteem Scale
Health services use	Emergency room (ER) visits, hospitalizations, premature institutionalization
Adverse effects	Psychological distress, increases in social isolation or loneliness
Indicators of social connections	
Social support	Duke-UNC Functional Social Support Questionnaire, Social support scale, social Provisions scale
Social engagement	Engagement in Meaningful Activities Survey (EMAS)
Social cohesion	The Group Cohesion Scale-Revised; Group Therapy Experience Scale, Group Environment Questionnaire, measures of neighborhood cohesion
Social capital	The World Bank's integrated questionnaire for the measurement of social capital (SC-IQ)
Cost and cost-effectiveness outcomes	
Cost-effectiveness	Cost-effectiveness analysis, cost utility analysis
Healthcare or social care utilization costs	Cost of service use
Cost per participant	Cost of service use per participant
Process indicators	
Acceptability	Various survey tools to measure acceptability
Adherence	Various survey tools to measure adherence
Feasibility	Various survey tools to measure feasibility
Barriers	e.g., language and cultural barriers, financial accessibility, hearing or vision impairments,
Increased awareness of community services	Various survey tools to measure awareness

8911803, 2024, 2, Downloaded

applicable Creative Commo

The reference lists of all included systematic reviews were not screened to identify additional studies due to lack of resources.

4.7 | Analysis and presentation

4.7.1 | Report structure

We followed the reporting structure of Campbell EGMs with the standard headings: abstract, plain language summary, background, methods, results, discussion, and conclusion.

This report includes the description of the study flow with included studies, excluded studies, and any studies awaiting assessment as well as the PRISMA study flow diagram. We also present the

conceptual framework, tables, and figures summarizing the distribution of included primary studies and systematic reviews across all the coding categories. These categories include study designs, publication status, quality of systematic reviews, types of interventions, types of outcomes, population characteristics, settings, geographic distribution.

The evidence and gap map has interventions as the row dimension and outcomes as the column dimension. We use bubbles of different sizes to represent included primary studies and systematic reviews and different colors to distinguish primary studies and methodological quality of systematic reviews. The filters used in the map are based on the number of included studies and coded information. See a sample of the map in Figure 2.

4.7.2 | Filters for presentation

Additional dimensions of interest used as filters include:

- 1. Study characteristics: the publication status of included studies, study design, countries where studies were conducted, World Bank classification by income (low income economies, lower-middle income economies, upper-middle income economies, high income economies), and WHO regions (African Region, Region of the Americas, South-East Asian Region, European Region, Eastern Mediterranean Region, Western Pacific Region), and setting (personal home, independent living/residential home, assisted living, long-term care/nursing home, orphanages, schools, workplaces, community centers, art gallery or museums, medical facilities, prisons), equity focus (study aimed at/focused on disadvantaged across any PROGRESS-Plus factors) and equity analysis (assessing any differences in effects (benefit or harm) across any PROGRESS-Plus factors).
- Intervention characteristics: focus (loneliness, social isolation, or both); format (group-based or one-on-one), sectors (clinical and population health, transport, housing, work, nutrition, environment, education, leisure: arts and entertainment, and spiritual care).
- 3. Population characteristics: age groups, and other sociodemographic factors as well as needs.



the PROGRESS-Plus acronym to describe sociodemographic inclusion factors associated with health inequities (O'Neill et al., 2014). PROGRESS-Plus stands for Place of residence (urban/rural), Race/ ethnicity/culture and language, Occupation, Gender or sex, Religion, Education, Socioeconomic status, Social capital, Plus: personal characteristics (e.g., age, disabilities), relationship features (e.g., exclusion from school, parent drug use), and time-dependent relationships (e.g., leaving the hospital, released from prison or other times when an individual might be temporarily disadvantaged).

We documented whether studies focused on populations who are at risk or experiencing barriers to health and social care or health inequities across any PROGRESS-Plus factors. For these studies, since interventions target different populations, we documented how potentially vulnerable populations, older people for instance, are defined and identified (e.g., using case finding, outreach, community-based programs, screening in primary care, through formal service network or agencies).

In addition, for each study, we assessed whether studies analyzed differential effects across PROGRESS factors for populations experiencing inequities. We also assessed subgroup analysis across additional ("Plus") factors known to be important for special populations, including age, disability, social frailty, literacy, living status, and health status.

4.7.3 | Dependency

Equity analysis

Hide Headers

C Fullscreen

We assessed equity following the same methods used in the evidence and gap map on digital interventions to reduce social isolation and loneliness in older adults (Welch et al., 2022). We used

View Records

We treated multiple reports of the same study as one study. A study with multiple outcomes and interventions was shown multiple times on the map (once for each outcome or intervention identified).





DRAFT MAP: In-Person Interventions for Reducing Social Isolation and Loneliness

• High quality reviews • Moderate quality reviews • Low quality reviews • Critically low quality reviews • Randomized controlled trials • Generated using v.2.2.3 of the EPPI-Mapper powered by EPPI Reviewer and created with 🖤 by the Digital Non-randomized studies Solution Foundry team.

-WILEY-C Campbell_ Collaboration

Primary studies were mapped regardless of whether they are included in multiple systematic reviews. Systematic reviews were mapped to interventions and outcomes based on their PICO question.

4.8 Data collection and analysis

4.8.1 | Screening and study selection

Pairs of reviewers used Machine learning text mining in Eppi-Reviewer web-based software program (Thomas et al., 2020) to screen titles and abstracts independently (EG, SD, EB, VB, TH, AW, AA, PD, JH, RD, SA, RI, LM, AAA, AJ, and FJ). We initially screened 10% of the titles and abstracts. The priority screening function developed a classifier based on the probability of inclusion determined from the preliminary screening results and presented the most likely studies to be included first. We manually screened all the articles to ensure all potentially eligible studies were captured for the full text screening stage which was also conducted by two reviewers independently.

We did not screen the reference lists of included systematic reviews as planned to identify additional studies that may have been missed in the database searches.

All screening was done following the eligibility criteria (see Appendix 3).

4.8.2 | Data extraction and management

We developed and pilot tested a data extraction code set in Eppi-Reviewer for data collection (see Supporting Information Appendix 4). We used a set of included studies for testing. The same studies were coded by all the reviewers and the coding was assessed for agreement. Any discrepancies were discussed, and description of the coding criteria was modified for clarity as necessary. After the pilot test, members of the team (EG, SD, EB, VB, TH, AW, AA, PD, JH, RD, SA, RI, LM, AAA, AJ, and FJ) individually extracted and coded data. Some non-English papers (including Chinese, Mandarin, Korean papers) were coded by CWY, RY, and TAMTM. Automation and text mining was not used for coding.

We coded the study characteristics (study design, publication status, methodological quality assessment of systematic reviews), categories and subcategories of interventions and other intervention characteristics (focus, sectors, goals, and risk factors targeted), outcome domains and subdomains, population characteristics (using PROGRESS-Plus acronym), settings, locations (by country, WHO region, and World Bank income classification).

We coded how populations were recruited and whether they were selected based on disadvantages across any PROGRESS-Plus factors.

We also coded whether differential analysis across any PROGRESS-Plus factors was conducted in the studies and systematic reviews to understand any equity issues. We did not contact authors of studies or systematic reviews for any missing information given the expected size of the map (over 300 studies).

4.8.3 | Tools for assessing risk of bias/study quality of included reviews

Pairs of reviewers used the AMSTAR 2 tool (Shea et al., 2017) to assess the quality of systematic reviews independently. Any disagreements were resolved by discussion. Primary studies were not assessed for risk of bias or methodological quality following guidance for evidence maps (Snilstveit 2016; White et al., 2020).

4.8.4 | Methods for mapping

We used the EPPI-Mapping tool (Digital Solution Foundry and EPPI_Centre, 2020) to develop the evidence and gap map. It is a tool to generate and visualize maps as an interactive HTML file on a web browser. The map is generated using exported data from EPPI-Reviewer, the software for managing and coding data.

5 | RESULTS

5.1 | Description of studies

5.1.1 | Results of the search

Our search retrieved 15,698 records from databases and 100 from other sources. We screened 9704 titles and abstracts after duplicates and other ineligible studies were removed by automation tools. We identified 1244 potentially eligible articles and retrieved full texts of 1183 records that were assessed and 513 were included. See Figure 3 for the PRISMA flow diagram.

5.1.2 | Excluded studies

A total of 670 studies were excluded during the full-text review phase due to an inappropriate aim (n = 327), intervention (n = 78), or study design (n = 192), and some as duplicates (n = 73) (Figure 3).

A list of key excluded studies can be found in the table for "Characteristics of excluded studies." Three studies (Lee, 2014; Roessler, 2012; Scazufca, 2022) were excluded as the aim was not to reduce either social isolation or loneliness. Lee (2014) assessed the effects of a recall program on depression and self-esteem. Roessler et al. (2012) examined the effects of a physical activity and group counseling intervention on the quality of life in woman with polycystic ovary syndrome. Scazufca et al. (2022) assessed the effects of a psychosocial collaborative care intervention on depression in older adults who were disadvantaged socioeconomically.





FIGURE 3 PRISMA flow diagram.

One study (Lin et al., 2020) was excluded as the intervention group received horticultural therapy through 3D virtual reality.

Four studies were excluded for an incorrect study design (Fokkema & van Tilburg, 2007; Low et al., 2015; Mächler et al., 2022; Purewal et al., 2017). Fokkema and van Tilburg (2007) was a literature review of intervention to reduce loneliness that did not report using a systematic search. Low et al. (2015) was a primary study with no control group that assessed the effect of the Lifestyle Engagement Activity Program (LEAP) in older adults in aged care. Purewal et al. (2017) was a systematic review of companion animal studies that looked at association as opposed to effectiveness. Mächler et al. (2022) was a primary study that investigated the views of patients following participation in a holistic care program which aims to increase social activity instead of measuring effectiveness.

5.1.3 Studies awaiting classification (if applicable)

We were unable to find the full texts of 63 articles that were potentially eligible. See references of Studies awaiting classification.

Synthesis of included studies 5.2

We included 513 articles that utilized in-person interventions aimed at reducing loneliness and/or social isolation. Of these articles, 421 were primary studies and 92 were reviews presented in the evidence and gap map. The primary studies included both randomized controlled trials (n = 310, 74%) and non-randomized studies (n = 111, 26%).

WILEY-Campbell Collaboration

Most of the publications for both the reviews and primary studies were completed (reviews: n = 86, 93%; primary studies: n = 350, 83%). Ongoing registered trials or protocols (reviews: n = 6, 7%; primary studies: n = 68, 16%) and conference abstracts (reviews: n = 0, 0%; primary studies: n = 3, 1%) were less common (Table 3).

Of the included publications, around half were published in the past 5 years (reviews: n = 58, 63%; primary studies: n = 216, 51%). More specifically, the highest number of reviews and primary studies were published in 2021 (reviews: n = 26, 28%; primary studies: n = 73,

17%) and 2020 (reviews: *n* = 14, 15%; primary studies: *n* = 51, 12%) (Figure 4).

5.2.1 | Interventions

To be included in this EGM, interventions had to be focused on addressing either loneliness (reviews: n = 33, 36%; primary studies: n = 272, 65%), social isolation (reviews n = 28, 30%; primary studies:

Characteristics of included articles		Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Study design	Randomized studies	N/A	310 (74)
	Non-randomized studies	N/A	111 (26)
Publication status	Completed	86 (93)	350 (83)
	On-going	6 (7)	68 (16)
	Conference abstracts	0 (0)	3 (1)
Intervention focus	Loneliness	33 (36)	272 (65)
	Social isolation	28 (30)	82 (19)
	Both social isolation and loneliness	31 (34)	67 (16)
Intervention format	One-on-one	51 (55)	103 (24)
	Group-based	74 (80)	335 (80)
	Both one-on-one and group based	43 (47)	23 (5)
Intervention delivery	Self-guided delivery	3 (3)	11 (3)
	Interpersonal delivery	72 (77)	284 (67)
	Community-based delivery	61 (67)	176 (42)
	Societal level delivery	1 (1)	2 (0.48)
	Multi-component/complex delivery	5 (5)	47 (11)
Outcomes	Health and psychosocial outcomes	91 (99)	417 (99)
	Indicators of social connections	44 (48)	99 (24)
	Cost and cost-effectiveness outcomes	3 (3)	6 (1)
	Process indicators	3 (3)	34 (8)
Most reported population socio-	Age	60 (65)	387 (92)
demographic characteristics	Gender or Sex	58 (63)	356 (85)
	Education	7 (8)	210 (50)
	Health condition	40 (43)	173 (41)
	Marital status	5 (5)	147 (35)
Most common countries	USA	46 (49)	113 (27)
	UK	32 (35)	35 (8)
	China	14 (15)	44 (10)
	Australia	15 (16)	31 (7)
	The Netherlands	17 (18)	26 (6)

TABLE 3 Characteristics of included studies.

73

51

47

35



FIGURE 4 Publication year of included studies.

n = 82, 19%) or both (review: n = 31, 34%; primary studies: n = 67. 16%).

The included interventions were delivered using a group format (reviews: *n* = 74, 80%; primary studies: *n* = 335, 80%) or one-on-one (reviews: n = 51, 55%; primary studies: n = 103, 24%). For articles that included both group and one-on-one formats (reviews: n = 43, 47%; primary studies: n = 23, 5%), both options were coded. Few articles did not specify the format (reviews: n = 10, 11%; primary studies: n = 6, 1%).

The included articles involved interventions delivered across various sectors including clinical and population health, transport, housing, work, nutrition, environment, education, leisure: arts and entertainment, and spiritual care. Clinical and population health was the most reported sector (reviews: n = 74, 80%; primary studies: n = 295, 70%), followed by leisure (reviews: n = 40, 43%; primary studies: n = 112, 27%) and education (reviews: n = 22, 24%; primary studies: *n* = 52, 12%). Environment (reviews: *n* = 1, 1%; primary studies: n = 6, 1%), housing (reviews: n = 2, 2%; primary studies: n = 5, 1%) and transport (reviews: n = 2, 2%; primary studies: n = 0, 0%) were the least reported sectors. For interventions delivered across multiple sectors, all relevant sectors were coded (Figure 5).

The interventions were organized using the following categories according to who was administering the intervention: self-delivery interventions (reviews: n = 3, 3%; primary studies: n = 11, 3%), interpersonal delivery interventions (reviews: n = 72, 78%; primary studies: *n* = 284, 67%), community-based delivery interventions (reviews: n = 61, 66%; primary studies: n = 176, 42%), and societal level delivery interventions (reviews: n = 1, 1%; primary studies: n = 2,

0.48%). Some interventions involved multiple delivery strategies and were coded as complex (reviews: n = 5, 5%; primary studies: n = 47, 11%) in addition to all relevant delivery strategies. For example, an intervention that combined cognitive behavioral therapy and social support group was coded as both an interpersonal delivery and a community-based delivery intervention, as well as a multicomponent/complex intervention (Figure 6).

Each intervention category included multiple subcategories. The few interventions that used a self-delivery mode were further categorized into two subcategories: self-guided interventions for changing maladaptive cognition (reviews: n = 3, 3%; primary studies: n = 6, 1%) and self-guided social skills training and psychoeducation (reviews: *n* = 0, 0%; primary studies: *n* = 5, 1%) (Figure 7).

Interpersonal delivery interventions had four subcategories: changing cognition, social skills training and psychoeducation, healthcare support, and social support. For both reviews and primary studies, changing cognition (e.g., cognitive behavioral therapy, psychotherapy) was the most reported (reviews: n = 52, 57%; primary studies: n = 151, 36%), while healthcare support was the least reported (reviews: n = 10, 11%; primary studies: n = 30, 7%). Some studies used multi-component interventions that involved multiple interpersonal delivery interventions (reviews: n = 40, 43%; primary studies: n = 32, 8%) so all relevant categories were coded (Figure 8).

Interventions with a community-based delivery mode were organized into five subcategories: group activities, support groups, neighborhood approaches, age-friendly communities, and volunteering. Group activities were most common for both reviews and



FIGURE 5 Intervention sector reported in included articles.



FIGURE 6 Intervention delivery method reported in included articles.

primary studies (reviews: n = 40, 43%, primary studies: n = 106, 25%). Interventions to promote age-friendly communities (reviews: n = 1, 1%, primary studies: n = 0, 0%) and volunteering interventions (reviews: n = 8, 9%, primary studies: n = 2, 0.48%) were the least common. Some studies used multi-component interventions that involved multiple community-based delivery interventions (reviews: n = 22, 25%; primary studies: n = 9, 2%) so all relevant categories were coded (Figure 9).

Societal-level delivery interventions focusing on policies and norms encompassed four subcategories: public health policy that promotes social cohesion and inclusion, public education and awareness of social relationships, policies on the design of neighborhoods, and funding relevant research. Evidence across all four subcategories was very sparse with no reviews or studies reporting on public education and awareness of social relationships or funding relevant research. Only one primary study reported on public health policy that promotes social cohesion and inclusion (n = 1, 0.24%) and one on policies on the design of neighborhoods as an intervention (n = 1, 0.24%). Only one review reported about policies on the design of neighborhoods (n = 1, 1%).



FIGURE 7 Self delivery interventions reported in included articles.



FIGURE 8 Interpersonal delivery interventions reported in included articles.



FIGURE 9 Community-based delivery interventions reported in included articles.

19 of 40

5.2.2 | Outcomes

The outcome framework consisted of four categories, each with a differing number of subcategories. These categories included: health and psychosocial outcomes, indicators of social connections, cost and cost-effectiveness outcomes, and process indicators or implementation outcomes. For studies that looked at more than one outcome, all relevant outcomes were coded. Most reviews and primary studies reported at least one health and psychosocial outcome (reviews: n = 91, 99%; primary studies: n = 417, 99%). Few reviews and primary studies reported cost and cost-effectiveness outcomes (reviews: n = 3, 3%; primary studies: n = 34, 8%) (Figure 10).

Health and psychosocial outcomes consisted of eight subcategories: loneliness, social isolation, social connectedness, quality of life/ well-being, anxiety/depression, self-esteem or self-efficacy, adverse effects, and health services use. Loneliness was the health and psychosocial outcome reported most often (reviews: n = 67, 73%; primary studies: n = 325, 77%), followed by anxiety/depression (reviews: n = 48, 52%; primary studies: n = 181, 43%), and quality of life/wellbeing (reviews: n = 42, 46%; primary studies: n = 179, 43%). Adverse effects (reviews: n = 2, 2%; primary studies: n = 7, 2%) and health services use (reviews: n = 4, 4%; primary studies: n = 16, 4%) were the least reported health and psychosocial outcomes (Figure 11).

Indicators of social connections consisted of the following subcategories: social support, social engagement, social cohesion, and social capital. Social support (reviews: n = 36, 39%; primary studies: n = 82, 19%) and social engagement (reviews: n = 14, 15%; primary studies: n = 20, 5%) were the most reported indicators of social connections. Social cohesion (reviews: n = 0, 0%; primary studies: n = 2, 0.48%) and social capital (reviews: n = 2, 2%; primary studies: n = 6, 1%) were reported in few articles (Figure 12).

Cost and cost-effectiveness outcomes category had three subcategories: cost-effectiveness, cost per participant, and healthcare or social care utilization costs. Cost effectiveness (reviews: n = 3, 3%; primary studies: n = 5, 1%) was the most assessed, while healthcare or social care utilization costs (reviews: n = 1, 1%; primary studies: n = 2, 0.48%) and cost per participant (reviews: n = 1, 1%; primary studies: n = 2, 0.48%) were assessed in very few articles (Figure 13).

Process indicators or implementation outcomes consisted of the following subcategories: acceptance, adherence, and feasibility. More primary studies than reviews looked at these process indicators, with only one review reported to have looked at adherence, feasibility, and increased awareness of community services (n = 1, 1%). Acceptance (n = 21, 5%), feasibility (n = 15, 4%), and adherence (n = 14, 3%) were most common among the primary studies (Figure 14).

5.3 | Risk of bias in included reviews

Review quality was assessed using the AMSTAR2 Quality Assessment tool. Most reviews were classified as either low (n = 19, 21%) or critically low (n = 63, 68%). High (n = 1, 1%) (Reichow et al., 2012) and moderate (n = 3, 3%) (Bee et al., 2014; Jarvis et al., 2020; Moore et al., 2018) quality reviews were uncommon. Review protocols (n = 6, 6%) were not assessed (Figure 15).

The critical domains that were most poorly met were domain 2 (not explicitly reporting that review methods were established before the conduct of the review in 64%) and domain 7 (not providing a list of excluded studies and their justification of exclusion in 83%).

5.4 | Additional dimensions

5.4.1 | Region

This EGM identified publications from 50 different countries across the globe. For the primary studies, the countries where the



FIGURE 10 Outcomes reported in included articles.



FIGURE 11 Health and psychosocial outcomes reported in included articles.



FIGURE 12 Indicators of social connections reported in included articles.

interventions were conducted were coded. For reviews, the countries specified for their included studies were coded. It should be noted that if a review included multiple studies conducted in the same country, that country was only coded once. If one primary study was included in multiple reviews the country was coded for each review.

The most common countries were the USA (reviews: n = 46, 50%; primary studies: n = 113, 27%), the UK (reviews: n = 32, 35%; primary studies: n = 35, 8%), China (reviews: n = 14, 15%; primary studies: n = 44, 10%), and Australia (reviews: n = 15, 16%; primary studies: n = 31, 7%) (Figure 16).

The location in which the interventions took place were further categorized using the World Health Organization (WHO) regions and the World Bank income classification. Most studies were conducted in the European Region (reviews: n = 59, 64%; primary studies: n = 138,

33%), followed by the Americas (reviews: n = 47, 51%; primary studies: n = 126, 30%), and then the Western Pacific Region (reviews: n = 42, 46%; primary studies: n = 124, 29%). Few studies were conducted in the African Region (reviews: n = 1, 1%; primary studies: n = 1, 0.24%). Over a quarter of the reviews included studies from multiple regions (reviews: n = 25, 27%) so all relevant regions were coded (Figure 17). Some reviews did not report the countries of their included studies and therefore the relevant WHO Region and World Bank Income Classification could not be determined (n = 15, 16%).

The World Bank Income Classification coding matches that of the WHO Region coding, as most studies were conducted in the European Region and the Americas, which consist mostly of high-income countries. Countries with high-income economies (reviews: n = 72, 78%; primary studies: n = 320, 76%) were reported most often, followed

21 of 40

WELCH ET AL.



FIGURE 13 Cost and cost-effectiveness outcomes reported in included articles.



FIGURE 14 Process indicators or implementation outcomes reported in included articles.

by upper-middle income countries (reviews: n = 28, 30%; primary studies: n = 67, 16%), then lower-middle income countries (reviews: n = 15, 16%; primary studies: n = 34, 8%). No studies took place in low-income countries. All relevant income classifications were coded for reviews that included primary studies from multiple countries with various income classes (reviews: n = 29, 31%) (Figure 18).

5.4.2 | Settings

Many articles did not specify the setting (reviews: n = 56, 61%; primary studies: n = 134, 32%) that the intervention was administered in. Of those that did specify the setting, community settings, such as a community center or a park, were the most common (reviews:

n = 24, 26%; primary studies: *n* = 90, 21%). Long-term care, or nursing homes were the second most reported setting (reviews: *n* = 23, 25%; primary studies: *n* = 53, 13%). Only one review reported the setting to be a prison (*n* = 1, 1%), and only 2 primary studies reported the setting to be an orphanage (*n* = 2, 0.48%). For studies involving interventions administered in multiple settings (reviews: *n* = 32, 34%; primary studies: *n* = 24, 6%), all relevant settings were coded (Table 4).

5.4.3 | Participants needs

The needs of the participants in the included studies were identified. Social and emotional needs were the most prevalent (reviews: n = 89,



FIGURE 15 Quality assessment of included reviews (n = 92).



FIGURE 16 Geographic heatmap of included primary studies.

97%, primary studies: n = 401, 95%), followed by learning a new activity or language (reviews: n = 14, 15%; primary studies: n = 42, 10%) and clinical/health needs (reviews: n = 5, 5%; primary studies: n = 55, 13%). No reviews or primary studies reported participant needs for respite care. For studies involving populations with multiple needs (reviews: n = 22, 34%; primary studies: n = 123, 29%), all relevant needs were coded (Table 5).

5.4.4 | Participant characteristics (PROGRESS-Plus)

We documented all PROGRESS-Plus factors described for participants in the included reviews and primary studies. Among the PROGRESS-Plus factors, age (reviews: n = 61, 65%; primary studies: n = 387, 92%) and gender or sex (reviews: n = 58, 63%; primary studies: n = 356, 85%) were the most reported. No reviews reported the religion of participants (Table 6).

Participant ages were arranged into groups based on life stages, such as childhood, adolescence, or middle-aged adulthood. Most reviews and primary studies included older adults between the ages of 60–75 (reviews: n = 55, 60%; primary studies: n = 226, 54%). Young adults aged 25–44 were the age group that was the least represented (reviews: n = 24, 26%; primary studies: n = 119, 28%) among adults. Among young people, adolescents aged 10–24 were the most represented (reviews: n = 25, 27%; primary studies: n = 135, 32%), while

23 of 40



FIGURE 17 World Health Organization regions reported in included articles.



FIGURE 18 World Bank income classification of countries in included articles.

children under the age of 10 were included in the least articles (reviews: n = 11, 12%; primary studies: n = 34, 8%). All age groups were represented and more than half of the studies involved multiple age groups (reviews: n = 63, 70%; primary studies: n = 263, 63%). Around a third of the reviews and some primary studies did not report the age of participants (reviews: n = 33,

35%; primary studies: n = 39, 9%) (Figure 19). Overall, more reviews and primary studies combined focused on older adults ≥ 60 years (n = 307, 60%) than young people ≤ 24 years (n = 176, 34%).

Of the studies involving children or adolescents (reviews: n = 25, 27%; primary studies: n = 151, 36%), no reviews and few

TABLE 4 Intervention setting reported in included articles.

Setting	Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Community center, park, etc.	24 (26)	90 (21)
Long-term care/nursing home	23 (25)	53 (13)
Personal home	20 (21)	55 (13)
Medical center	6 (7)	37 (9)
School	11 (12)	48 (11)
Independent living in residential home	5 (5)	11 (3)
Assisted living	4 (4)	7 (2)
Art gallery or museum	2 (2)	3 (1)
Workplace	0 (0)	9 (2)
Prison	1 (1)	0 (0)
Orphanage	0 (0)	2 (0.48)

TABLE 5 Needs of participants reported in included articles.

Needs	Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Social and emotional needs	89 (97)	401 (95)
Learning (e.g., a new activity, language)	14 (15)	42 (10)
Clinical health needs	5 (5)	54 (13)
Skills development	3 (3)	15 (4)
Purpose in life	2 (2)	15 (4)
Personal care needs	0 (0)	8 (2)
Meals	3 (3)	4 (1)
Accommodation (e.g., housing/ home modifications and maintenance	1 (1)	6 (1)
Mobility	3 (3)	3 (1)
Communication (e.g., language support/interpreters)	0 (0)	3 (1)
Domestic assistance	0 (0)	3 (1)
Financial management	0 (0)	2 (0)
Care navigation support or task orientation	1 (1)	6 (1)
Caregiver support	2 (2)	6 (1)
Respite care	0 (0)	0 (0)

primary studies reported parental status (reviews: n = 0, 0%; primary studies: n = 12, 8%). Less than 5% of the primary studies reported children to have single parents (n = 6, 4%) or working parents (n = 6, 4%).



TABLE 6	Population sociodemographic characteristics reported
in included	articles.

Population sociodemographic characteristics	Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Place of residence	5 (5)	49 (12)
Race/ethnicity	9 (10)	105 (25)
Occupation	2 (2)	83 (20)
Gender or sex	58 (63)	356 (85)
Religion	O (O)	27 (6)
Education	7 (8)	210 (50)
Socioeconomic status	5 (5)	96 (23)
Marital status	5 (5)	147 (35)
Living alone	4 (4)	66 (16)
Age groups	61 (65)	387 (92)
Health condition	40 (43)	173 (41)
Caregivers	7 (8)	10 (2)

5.4.5 | Health condition of participants

Participants were reported as having specific health conditions in 226 articles (reviews: *n* = 41, 44%; primary studies: *n* = 185, 44%). Since depression is commonly associated with social isolation and loneliness, we presented it separate from other mental disorders. For reviews, mental health disorders such as schizophrenia, psychosis, bipolar disorder, neuropsychiatric disorders, attention deficit hyperactivity disorder (ADHD), oppositional defiant disorder (ODD), or post-traumatic stress disorder (PTSD), but not including depression, were most reported (n = 14, 15%), followed by non-communicable diseases such as cardiovascular diseases, diabetes, or cancer (n = 12, 13%). No reviews reported participants with communicable diseases, psychological or social frailty, care dependence, discharge from hospital, or those in endof-life/palliative care. For primary studies, non-communicable diseases were most reported (n = 63, 15%), followed by depression (n = 42, 10%). Discharge from the hospital and end-of-life or palliative care were reported in one primary study each (n = 1, n)0.24%). All conditions were coded for studies involving participants with multiple conditions (reviews: n = 24, 26%; primary studies: n = 40, 16%) (Table 7).

5.4.6 | Equity analysis

All but two reviews and 14 primary studies reported an equity focus on populations disadvantaged across PROGRESS-Plus. The most frequently reported focus was age with nearly 60% of reviews and primary studies focused on age (reviews: n = 53, 57%; primary studies: n = 247, 59%). The next most reported equity



FIGURE 19 Age of participants reported in included articles.

TABLE 7 Health conditions of participants reported in included articles.

Health condition	Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Communicable disease	0 (0)	5 (1)
Non-communicable disease	12 (13)	63 (15)
Autism, Asperger syndrome, ADHD	7 (8)	10 (2)
Depression	11 (12)	42 (10)
Dementia	11 (12)	10 (2)
Mental health disorders	14 (15)	45 (11)
Comorbidity	3 (3)	14 (3)
Disability	4 (4)	14 (3)
Physical frailty	3 (3)	7 (2)
Psychological frailty	0 (0)	6 (1)
Social frailty	O (O)	4 (1)
Care dependent	O (O)	7 (2)
Discharge from hospital	0 (0)	1 (0.24)
End-of-life/palliative care	0 (0)	1 (0.24)

foci were loneliness (reviews: n = 40, 43%; primary studies: n = 204, 49%), health status (reviews: n = 20, 22%; primary studies: n = 116, 27%), social isolation (reviews: n = 32, 34%; primary studies: n = 52, 12%), and living situation (reviews: n = 16,

17%; primary studies: n = 69, 16%), respectively. Religion was the least reported focus, with only one primary study reporting it as a focus (n = 1, 0.24%). Two thirds of the articles focused on populations disadvantaged across multiple progress factors (reviews: n = 59, 63%; primary studies: n = 281, 67%) (Table 8).

Included articles used various strategies to identify participants at risk for social isolation or loneliness. Case-finding was the most common strategy for reviews (n = 11, 12%), while outreach was the most common for primary studies (n = 90, 21%), followed by case-finding (n = 77, 18%). Some of the articles utilized multiple strategies to recruit participants that were disadvantaged (reviews: n = 8, 9%; primary studies: n = 34, 8%). Most reviews and a quarter of the primary studies did not specify a strategy (reviews: n = 70, 76%; primary study: n = 105, 25%) (Figure 20).

Though many articles reported an equity focus, few conducted an analysis of the effects across PROGRESS-Plus. More primary studies than reviews performed a subgroup or regression analysis. Of the primary studies that performed an analysis, analysis by gender or sex was the most common (n = 13, 3%), followed by health status (n = 9, 2%), social capital (n = 7, 2%), and age (n = 5, 1%). For the reviews, analysis by age and health status (both n = 2, 2%) were most common. No reviews or primary studies performed an analysis to assess differences in effects across occupation or education. Few articles conducted an analysis of the intervention effectiveness across multiple PROGRESS-Plus factors (reviews: n = 1, 1%; primary studies: n = 10, 2%) (Table 9).

6 | DISCUSSION

6.1 | Summary of main results

This evidence and gap map has 513 included articles, majority of which are primary studies (n = 421), and the remaining of which are reviews (n = 92). Most articles were conducted in high-income

TABLE 8 Equity focus across PROGRESS-Plus factors reported in included articles.

PROGRESS-Plus factors	Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Place of residence	0 (0)	21 (5)
Race/ethnicity	2 (2)	18 (4)
Occupation	1 (1)	13 (3)
Gender/sex	2 (2)	36 (9)
Religion	0 (0)	1 (0.24)
Education	1 (1)	7 (2)
Socioeconomic status	3 (3)	11 (3)
Social capital	1 (1)	15 (4)
Plus factor—Socially isolated or at risk	32 (34)	52 (12)
Plus factor—Lonely or at risk	40 (43)	204 (49)
Plus factor—Age	53 (57)	247 (59)
Plus factor-Disability	11 (12)	31 (7)
Plus factor—Frailty	0 (0)	5 (1)
Plus factor—Health status	20 (22)	115 (27)
Plus factor—Living situation	16 (17)	69 (16)

Campbell_WILEY

27 of 40

countries in the European region, the Americas or the Western Pacific region. The vast majority were published between 2018 and 2022, with over 60% of reviews and 50% of primary studies having been completed in this period. However, many of these reviews are of critically low or low quality.

There is an uneven distribution of the evidence across the intervention and outcome mapping framework, with a similar pattern

TABLE 9	Equity analyses across	PROGRESS-Plus	factors
conducted in	included articles.		

PROGRESS-Plus factors	Number of reviews (%) (n = 92)	Number of primary studies (%) (n = 421)
Place of residence	1 (1)	0 (0)
Race/ethnicity	1(1)	2 (0.48)
Occupation	0 (0)	0 (0)
Gender/sex	1 (1)	13 (3)
Religion	0 (0)	1 (0.24)
Education	0 (0)	0 (0)
Socioeconomic status	0 (0)	2 (0.48)
Social capital	0 (0)	7 (2)
Plus factor—Socially isolated or at risk	0 (0)	3 (1)
Plus factor—Lonely or at risk	0 (0)	2 (0.48)
Plus factor—Age	2 (2)	5 (1)
Plus factor-Disability	0 (0)	1 (0.24)
Plus factor—Frailty	0 (0)	1 (0.24)
Plus factor—Health status	2 (2)	9 (2)
Plus factor—Living situation	1 (1)	1 (0.24)



FIGURE 20 Recruitment strategies for at-risk populations reported in included articles.

-WILEY-C Campbell Collaboration

of evidence and gaps seen between the reviews and primary studies. Most reviews and primary studies assessed interpersonal delivery and community-based delivery interventions. Interventions for changing cognition and group activities were the most common.

Health and psychosocial outcomes were the most assessed, especially loneliness, anxiety/depression, and quality of life. Social support and social engagement were the most assessed indicators of social connections.

All age groups were represented but more studies and reviews disproportionately focused on older adults ≥60 years (60%) than young people ≤24 years (34%). Although many reviews and primary studies had an equity focus on populations disadvantaged across PROGRESS-Plus factors, only few focused on reaching vulnerable populations across the PROGRESS factors. Most focused on reaching vulnerable populations across the Plus factors, such as populations at risk of social isolation or loneliness, age, health status, and living situation. However, few described how at-risk populations were recruited and even fewer performed an analysis to assess differences in effects across PROGRESS-Plus factors for populations experiencing inequities.

People using this map will be able to find all systematic reviews and primary studies of the interventions of interest. However, they need to be aware that systematic reviews and primary studies with multiple interventions or intervention components are coded as relevant for each intervention.

6.2 | Areas of major gaps in the evidence

Major gaps were identified in societal level delivery interventions focusing on norms, laws, and policies (e.g., public health policies that promote cohesion and inclusion). Community-based delivery interventions focusing on structural changes in the community to improve opportunities for social interactions or access to amenities (e.g., age-friendly communities with well-maintained sidewalks, stairs that have railings, recreational facilities, accessibility ramps) were lacking. Some outcomes were hardly assessed such as some indicators of social connections (social capital and social cohesion), cost and cost-effectiveness, and barriers of implementation or access of interventions. The quality assessment with AMSTAR2 showed a major gap in high or moderate quality reviews.

Social isolation and loneliness are major public health concerns that can be addressed through collaboration with other sectors. Most of the evidence is from the clinical and population health sector with major gaps in other sectors including transportation, housing, work, nutrition, environment, and spiritual care indicating the need for more collaboration across sectors to effectively alleviate social isolation and loneliness.

There were gaps in the geographic distribution of studies with limited evidence available for Africa and none in low-income countries.

6.3 | Potential biases in the mapping process

The search strategy for identification of potentially eligible studies was comprehensive and covered published literature until February

2022. Rigorous methods were used to identify eligible studies and reviews, and to assess review quality, including screening title and abstract, and full text for inclusion in duplicate, as well as assessment of review quality using AMSTAR2 in duplicate, with all conflicts resolved via consensus. However, we did not assess the quality of primary studies and the certainty of the evidence is therefore inconclusive.

The research question was on the effectiveness of in-person faceto-face interventions for reducing social isolation and loneliness, and we only included study designs which assessed effectiveness. Outcomes were not used as eligibility criteria, only the study population, intervention, study aim and study design. Studies assessing the effects of in-person face-to-face interventions on social isolation and loneliness and other indicators of social connections such as quality of life or depression/anxiety were included.

Our search may not have captured all gray literature or unpublished and more recently published eligible studies or reviews. No language restrictions were applied to the search, but mainly English-language databases were searched, and it is possible some relevant studies published in non-English languages may have been missed. Furthermore, reference lists of included systematic reviews were not screened, and therefore some additional eligible studies may not have been identified. In addition, we did not contact authors for missing information which may limit the mapping of evidence.

Lastly, mapping the interventions into the mutually exclusive categories of the conceptual framework developed for this evidence and gap map was challenging due to the lack of consistency between articles with respect to terminology, as well as lack of a standardized framework for the classification of interventions for social isolation and loneliness. The socio-ecological framework has been adopted in most existing frameworks and reviews, but it does not allow for mutually exclusive categorization of interventions since it examines risk and protective factors that occur and interact across levels. We therefore developed a pragmatic approach to categorize interventions focusing on who delivered the interventions and where they were delivered.

6.4 | Limitations of the EGM

This evidence and gap map does not include primary studies without a control group, nor studies with a qualitative design or association studies. While these studies may provide valuable information about interventions to reduce social isolation and loneliness, they do not assess the effectiveness of these interventions.

Most included articles do not focus on reaching populations vulnerable across PROGRESS-Plus factors, such as place of residence (urban/rural), race/ethnicity (ethnic minority groups), gender/sex (LGBTQIA2S+ populations), religion, socioeconomic status, living alone and caregivers. While focusing on addressing the needs of those at risk or experiencing social isolation and/or loneliness, the elderly, or children, those with a disability or health condition, there are other vulnerable populations who are disadvantaged across the

PROGRESS factors that may also be socially isolated or lonely and could benefit from targeted interventions.

Less than 10% of all included articles assessed process indicators or implementation outcomes, which means there is limited evidence regarding barriers to implementation of these interventions, as well as their acceptability, feasibility, and participant adherence.

In addition, only two non-randomized studies and one moderate quality review assessed societal level delivery intervention, thus there is limited evidence regarding the effectiveness of these interventions. Although many studies and reviews assessed community-based delivery interventions, there is limited evidence of the effectiveness of interventions focusing on structural community changes to reduce social isolation or loneliness since only one review considered interventions to promote age-friendly communities.

A higher proportion of included reviews and studies focused on older adults than young people. Social isolation and loneliness have been studied more in adults than young people (Eccles et al., 2021) and most of the existing frameworks we found focused on the experiences of older adults; only one focused on children. There is therefore limited information about differences that might exist between older adults and younger people's experience of social isolation and loneliness.

6.5 | Stakeholder engagement throughout the EGM process

We sought input from stakeholders from organizations interested in social isolation and loneliness as described in the methods section (Stakeholder engagement)and there were no changes with stakeholder engagement throughout the EGM process.

7 | AUTHORS' CONCLUSIONS

7.1 | Implications for research, practice and/or policy

Social isolation and loneliness are growing public health concerns, which have been highlighted by the COVID-19 pandemic with respect to lockdowns, as well as by the surge of research appearing in more recent years. This evidence and gap map shows the available evidence regarding in-person interventions to alleviate social isolation or loneliness within any age group. Of the 92 systematic reviews that were included in this map, nearly 70% were of critically low quality, and over 40% have been published since 2020, with most included reviews being published in 2021. Much of the evidence is clustered around interpersonal delivery and community-based delivery interventions, which reported on health and psychosocial outcomes or indicators of social connectedness. Most of these interventions focused on an individual perspective by reducing loneliness or anxiety/depression symptoms, and/or improving

participant quality of life/well-being. Most included articles focused on reaching vulnerable populations across the Plus factors, such as populations at risk of social isolation or loneliness, age, health status, and living situation.

Major gaps in the evidence can be seen in research that looks at interventions based on structural community changes (e.g., age-friendly communities with well-maintained sidewalks, stairs that have railings, recreational facilities, accessibility ramps) and societal-level changes. There were gaps in research assessing process indicators for outcomes, such as acceptance, adherence, barriers, and feasibility of intervention implementation. With the rise in levels of social isolation and loneliness (Ernst et al., 2022; O'Sullivan et al., 2021), it is important to not only consider the efficacy of interventions which aim to alleviate social isolation and loneliness, but also the feasibility and barriers to implementing these interventions, as well as their acceptability by participants. Very few reviews and primary studies described how atrisk populations were recruited and even fewer performed an analysis to assess differences in effects across PROGRESS-Plus factors for populations experiencing inequities. Moreover, much of the research was done in high-income or upper-middle-income countries, specifically in the United States, United Kingdom, China, Australia, and the Netherlands, with no research conducted in low-income countries. Loneliness and social isolations are common issues but complex to resolve-we need whole systems solutions and therefore interventions that only utilize one type of approach may be effective only in a specific population/community. To combat loneliness and social isolation more effectively, we need effective strategies and solutions across all sectors working in synergy with each other.

Despite recent increases in evidence in the past years, there is an uneven distribution, with most focusing on interpersonal delivery or community-based delivery interventions which aim to alleviate loneliness or symptoms of depression/anxiety or improve participant quality of life/well-being. The majority of published systematic reviews are of critically low quality, emphasizing the need for more high-quality reviews.

This evidence and gap map contributes toward building an evidence base that can be used to inform future policy decisions, as well as guide future researchers and funders to fill in some of the identified evidence gaps.

ACKNOWLEDGMENTS

We gratefully acknowledge the participation of stakeholders in our stakeholder consultation.

This project was funded by the World Health Organization.

CONTRIBUTIONS OF AUTHORS

- Content: VW, ETG, NH, SB, SD, EdG, JHL, ML, MLim, KM, PQ, CC, PH, CM
- EGM methods: VW, ETG, PH, CM, SD, VB, AA, EB, MB, PD, RD, JH, TH, AJ, FJ, LM, WYC, RMY, TAMTM, AW, AAA, RI, SA
- Information retrieval: DS

DECLARATIONS OF INTEREST

Vivian Welch is editor-in-chief of the Campbell Collaboration. The editorial process was handled by an independent editor and VW had no input in the editorial process or decisions.

Julianne Holt-Lunstad and Michelle Lim were co-investigators on the KIND challenge intervention aimed at reducing loneliness which was sponsored by the app Nextdoor.

Dr. Kate Mulligan has been a contract consultant for the Canadian Red Cross, a provider of in-person interventions to reduce social isolation and loneliness. She has also been involved in intervention projects related to social prescribing with the Alliance for Healthier Communities.

Pamela Qualter was involved in two reviews used to gather data for the current evidence gap review; she was not involved in any primary research in the subject area of the review.

Mathias Lasgaard was involved in one review used to gather data for the current evidence gap review.

Sierra Dowling is the managing editor of the Campbell Ageing Coordinating Group, but the editorial process was handled by another managing editor.

Paul Hebert is the Chief Medical and Science advisor for the Canadian Red Cross.

Christopher Mikton works for the World Health Organization which has helped fund this evidence and gap map.

Elizabeth Ghogomu, Wan Yuen Choo, Raudah Mohd Yunus, Tengku AM Tengku Mohd, Niobe Haitas, Sivan Bomze, Simone Dahrouge, Edward Garcia, Victoria Barbeau, Tarannum Hussain, Ella Beveridge, Arpana Wadhwani, Ali Al-Zubaidi, Payaam Desai, Jimmy Huang, Rebecca Doyle, Sarah Allam, Rayan Ibrahim, Leen Madani, Abdulah Al Ameer, Alyssa Jearvis, Mostafa Bondok, Fatima Jahel and Douglas M. Salzwedel, have no conflicts of interest.

PLANS FOR UPDATING THE EGM

The EGM will be updated every 2 years.

SOURCES OF SUPPORT

Internal sources

• None, Other

External sources

• World Health Organization, Switzerland

Purchase Order Number: 202759417

DIFFERENCES BETWEEN PROTOCOL AND REVIEW

We did not screen the references of included systematic reviews to identify additional primary studies because of time constraints.

CHARACTERISTICS OF STUDIES

Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Fokkema et al., 2007	Study design (not a systematic review of effectiveness)
Lee, 2014	Evidence/aim (not to reduce social isolation or loneliness)
Lin et al., 2020	Intervention (digital)
Low et al., 2015	Study design (a non-comparative primary study)
Mächler et al., 2022	Study design (a non-comparative primary study)
Purewal, 2017	Study design (not a systematic review of effectiveness)
Roessler, 2012	Evidence/aim (not to reduce social isolation or loneliness)
Scazufca, 2022	Evidence/aim (not to reduce social isolation or loneliness)

AFFILIATIONS

¹Methods Centre, Bruyère Research Institute, Ottawa, Canada

²Methods Centre, Bruyère Research Institute, University of Ottawa, Ottawa, Canada

³School of Medicine, University College Cork, Cork, Ireland

⁴Faculty of Social Sciences, University of Ottawa, Ottawa, Canada

⁵Faculty of Medicine, University of British Columbia, Vancouver, Canada

⁶Faculty of Science, University of Ottawa, Charlottetown, Canada

⁷Faculty of Health Sciences, University of Ottawa, Ottawa, Canada

⁸Department of Social & Preventive Medicine, Faculty of Medicine, University of Malaya, Kuala Lumpur, Malaysia

⁹Public Health Medicine, University Teknologi MARA, Sungai Buloh, Malaysia

¹⁰Faculty of Medicine and Health Sciences, University Sains Islam Malaysia, Putra Nilai, Malaysia

¹¹Health Sciences, McMaster University, Hamilton, Canada

¹²Canadian Red Cross, Montreal, Canada

¹³Canadian Red Cross, Mississauga, Canada

¹⁴C. T. Lamone Primary Healthcare Research Centre, Bruyere Research Institute, Ottawa, Canada

 $^{15}\mbox{The}$ Coalition to End Social Isolation and Loneliness, Washington, District of Columbia, USA

¹⁶Department of Psychology, Brigham Young University, Provo, Utah, USA

¹⁷DEFACTUM, Public Health and Health Services Research, Central Denmark Region, Aarhus N, Denmark

¹⁸Iverson Health Innovation Research Institute, Swinburne University of Technology, Melbourne, Australia

¹⁹Ending Loneliness Together, Melbourne, Australia

²⁰Dalla Lana School of Public Health, University of Toronto, Toronto, Canada

²¹Department of Anesthesiology, Pharmacology and Therapeutics, University of British Columbia, Vancouver, Canada ²²Manchester Institute of Education, University of Manchester, Manchester, UK

²³Bruyère Research Institute, Palliative Care, University of Ottawa, Ottawa, Canada

²⁴Department of Social Determinants of Health, Division of Healthier Populations, World Health Organization, Geneva, Switzerland

REFERENCES

REFERENCES TO STUDIES EXCLUDED FROM THIS REVIEW

Fokkema et al., 2007 {published data only}

* Fokkema, C. M., & Van Tilburg, T. G. (2007). Loneliness interventions among older adults: Sense or nonsense. *Tijdschrift Voor Gerontologie En Geriatrie*, 38(4), 185–203.

Lee 2014 {published data only}

* Lee, Y.-S. (2014). The effect of individual reminiscence program on depression and self-esteem for female elders living alone (Vol. 8). JEJU National University Repository: dissertation. https://oak.jejunu.ac. kr/handle/2020.oak/21156

Lin et al., 2020 {published data only}

* Lin, T. Y., Huang, C. M., Hsu, H. P., Liao, J. Y., Cheng, V. Y. W., Wang, S. W., & Guo, J. L. (2020). Effects of a combination of threedimensional virtual reality and hands-on horticultural therapy on institutionalized older adults' physical and mental health: Quasiexperimental design. *Journal of Medical Internet Research*, 22(11), e19002.

Low et al., 2015 {published data only}

* Low, L. F., Baker, J. R., Harrison, F., Jeon, Y. H., Haertsch, M., Camp, C., & Skropeta, M. (2015). The Lifestyle Engagement Activity Program (LEAP): Implementing social and recreational activity into casemanaged home care. *Journal of the American Medical Directors Association*, 16(12), 1069–1076.

Mächler et al., 2022 {published data only}

* Mächler, R., Sturm, N., Frick, E., Schalhorn, F., Stolz, R., Valentini, J., & Straßner, C. (2022). Evaluation of a spiritual history with elderly multi-morbid patients in general practice—A mixed-methods study within the project HoPES3. International Journal of Environmental Research and Public Health, 19(1), 538.

Purewal 2017 {published data only}

* Purewal, R., Christley, R., Kordas, K., Joinson, C., Meints, K., Gee, N., & Westgarth, C. (2017). Companion animals and child/adolescent development: A systematic review of the evidence. International Journal of Environmental Research and Public Health, 14(3), 234.

Campbell_WILEY

31 of 40

Roessler 2012 {published data only}

* Roessler, K. K., Glintborg, D., Ravn, P., Birkebaek, C., & Andersen, M. (2012). Supportive relationships- psychological effects of group counselling in women with polycystic ovary syndrome (PCOS). *Communication & Medicine*, 9(2), 125.

Scazufca 2022 {published data only}

* Scazufca, M., Nakamura, C. A., Seward, N., Moreno-Agostino, D., van de Ven, P., Hollingworth, W., & Araya, R. (2022). A task-shared, collaborative care psychosocial intervention for improving depressive symptomatology among older adults in a socioeconomically deprived area of Brazil (PROACTIVE): A pragmatic, two-arm, parallel-group, cluster-randomised controlled trial. *The Lancet Healthy Longevity*, 3(10), e690–e702.

REFERENCES TO STUDIES AWAITING ASSESSMENT

Andersson 2015 {published data only}

Andersson, E., Christensson, K., & Hildingsson, I. (2015). Comparison of parental stress six months after birth in parents receiving group based antenatal care or standard antenatal care-clinical trial. Archives of Women's Mental Health, 18(2), 296–297.

Arnetz 1985 {published data only}

Arnetz, B. B. (1985). Gerontic occupational therapy—Psychological and social predictors of participation and therapeutic benefits. *American Journal of Occupational Therapy*, 39(7), 460–465.

Bergeest 1977 {published data only}

Bergeest, H. G., Steinbach, I., & Tausch, A. M. (1977). Psychic help for aged people of day-care-centers by attending person-centered encounter groups (author's transl). *Aktuelle Gerontologie*, 7(6), 305-313.

Bikomeye 2021 {published data only}

Bikomeye Jean, C., & Kwarteng Jamila, L. (2021). Social isolation and cancer mortality among African American women in the United States: A systematic review. Annals of Behavioral Medicine, 55, S423.

Campise 1988 {published data only}

Campise, R. L. (1988). The efficacy of social skills training and cognitive restructuring in alleviating loneliness [Doctoral thesis]. University of Kansas. ProQuest Digital Dissertations. p. 154.

Cecen 2007 {published data only}

Cecen, A. R., & Kocak, E. (2007). An experimental study: The effect of self esteem enhancement programme on middle school students' self esteem level. Egitim Arastirmalari-Eurasian Journal of Educational Research, 7(27), 59–68.

Cevizci 2009 {published data only}

Cevizci, S., Erginoz, E., & Baltas, Z. (2009). An assisted therapy concept for improving of mental health: Animal assisted therapy. *Nobel Medicus*, 5(1), 4–9 (in Turkish).

Chung 2003 {published data only}

Chung, H., & Kim, M. (2003). The effects of cognitive-behavioral group counseling on elderly depression, loneliness, and dysfunctional attitude. *Korean Journal Of Counseling And Psychotherapy*, 15(3), 477-490.

Clarke 2004 {published data only}

Clarke, A. M. (2004). The influence of a neighbourhood walking program on perceptions of social isolation among older adults. *Journal of Aging and Physical Activity*, 12(3), 401.

Cummings 1987 {published data only}

Cummings, L. (1987). Developmental treatment of childhood social isolation. *Dissertation Abstracts International*, 47(10-B), 4320.

Dahm 2009 {published data only}

Dahm, K. T., Landmark, B., & Reinar, L. M. (2009). The importance of personnel competence to achieve social participation and activity amongst users of municipal home care. NIPH Systematic Reviews: Executive Summaries. Knowledge Centre for the Health Services at The Norwegian Institute of Public Health (NIPH).

Fox 1998 {published data only}

Fox, P. G., Cowell, J. M., Montgomery, A. C., & Willgerodt, M. A. (1998). Southeast Asian refugee women and depression: A nursing intervention. *International Journal of Psychiatric Nursing Research*, 4(1), 423–432.

Gallant 1997 {published data only}

Gallant, W., Holosko, M., Gorey Kevin, M., & Lesiuk Teresa, L. (1997). Music as a form of intervention with out-patient alcoholic couples: A quasi-experimental investigation. *Canadian Journal of Music Therapy*, 5(1), 67–84.

Hames 2009 {published data only}

Hames, A., & Rollings, C. (2009). A group for the parents and carers of children with severe intellectual disabilities and challenging behaviour. *Educational and Child Psychology*, 26(4), 47–54.

Hang 2015 {published data only}

Hang, R.-H., Tao, J.-H., & Wu, M.-F. (2015). Effect of counseling on depression symptoms and loneliness of bereaved parents. *Chinese Journal of Clinical Psychology*, 23(2), 369–371.

Hare-Duke 2020 {published data only}

Hare-Duke, L. (2020). Adaptation and evaluation of an intervention to reduce loneliness amongst adults with depression [Doctoral dissertation, University of Nottingham]. http://eprints.nottingham.ac.uk/59651/

Hausman 1983 {published data only}

Hausman, M. S (1983). A comparison of cognitive behavioral and social skill approaches in the treatment of loneliness [Doctoral dissertation, Southern Illinois University at Carbondale]. https://www.elibrary.ru/ item.asp?id=7418914

He 2005 {published data only}

He, J. Y., & Long, R. F. (2005). Effects of social occupation technique exercises on the social functional recovery of rural schizophrenic patients. *Chinese Journal of Clinical Rehabilitation*, 9(8), 48–50.

Ho 2012 {published data only}

Ho, Y., Cheng, C. S. Y., & Chau Janita, P. C. (2012). The effectiveness of caregiver psychosocial interventions on the psychosocial wellbeing, physical health and quality of life of stroke family caregivers and their stroke survivors: A systematic review. JBI Library of Systematic Reviews, 10(12), 679–797. https://doi.org/10.11124/jbisrir-2012-66

Hopman-Rock 2002 {published data only}

Hopman-Rock, M., & Westhoff, M. H. (2002). Health education and exercise stimulation for older people: Development and evaluation of the program "Healthy and Vital". *Tijdschrift Voor Gerontologie En Geriatrie*, *33*(2), 56–63.

Jaboor 2011 {published data only}

Jaboor, Z. (2011). Social connection and mental health: a review of the literature for young people in rural communities...Occupational Therapy Australia, 24th National Conference and Exhibition, 29 June
1 July 2011. Australian Occupational Therapy Journal, 58, 25. https://doi.org/10.1111/j.1440-1630.2011.00937.x

Jarczok 2012 {published data only}

Jarczok, M., Jarczok, M. N., Obermann, K., Glazinski, B., Besier, W., & Mayer, M. (2012). A key to the heart of CHD patients? A randomized controlled trial study protocol. *Psychosomatic Medicine*, 74(3), A42.

Jingna 2012 {published data only}

Jingna, Y., Liqun, C., & Shoumei, J. (2012). Influence of community nurse-led multidisciplinary team home visiting services on psychological state of old age of home elderly. *Chinese Nursing Research*, 26(4B), 975–978.

Kim 2020 {published data only}

Kim, J. S., Lim, Y. M., & Lim, J. H. (2020). Effect of cognitive behavior group intervention in loneliness, depression, and suicidal ideation among the community-dwelling older adults with a high risk of depression. *The Journal of Korean Nursing Research*, *3*(2), 25–35. https://doi.org/10.34089/jknr.2019.3.2.25

Klein 2021 {published data only}

Klein, M., Kemper, N., Lich, K., & Winter-Wilms, F. (2021). Older, lonely men with alcohol dependence: Systematic research review concerning the current state of age and gender specific interventions. *Suchtherapie*, 22(02), 75–85. https://doi.org/10.1055/a-1419-9514

Lamb 2010 {published data only}

Lamb, D. C. W. (2010). An evaluation of an aerobic exercise intervention on the psychological health of children at risk of social exclusion [Doctoral dissertation. University of Nottingham].

Lee 2004 {published data only}

Lee, H., Jang, M.-J., & Kim, S.-M. (2004). The effect of friendship improvement program on the senses of alienation and social skills of middle school students. *Korean Journal of Youth Studies*, 11(3), 181–198.

Longabaugh 1978 {published data only}

Longabaugh, R., Fowler, D. R., Hostetler, M., McMahon, L., & Sullivan, C. (1978). Focus on patient problems: use of the problem-oriented record in a proposed evaluation study of social isolation. QRB. Quality Review Bulletin, 4(4), 4–7.

Melville 2020 {published data only}

Melville Nancy, A. (2020). Shared medical appointments educate and encourage patients with MS: The group dynamic addresses social isolation and reduces emergency room utilization. *Neurology Reviews*, 28(4), 34–37.

Odaci 2003 {published data only}

Odaci, H. (2003). The effect of human relations skills training upon loneliness levels of university students. *Psikiyatri Psikoloji Psikofarmakoloji Dergisi*, 11(4), 310–315.

Osberg 1982 {published data only}

Osberg Timothy, M. (1982). Social skills training with children: An update. Child Study Journal, 12(1), 57–75.

Panton 1979 {published data only}

Panton, J. H. (1979). MMPI profiles associated with outcomes of intensive psychotherapeutic counseling with youthful first offender prison inmates. *Research Communications in Psychology*, *Psychiatry and Behavior*, 4(4), 383–396.

Campbell_WILEY-

Patterson 2009 {published data only}

Patterson, C., & Lewin, G. (2009). Results from a small randomised controlled trial on a volunteer-led model of social enablement for older people on measures of depression, loneliness, and wellbeing. *Australasian Journal on Ageing*, 28, A70.

Pitkala 2005 {published data only}

Pitkala, K., Routasalo, P., & Tilvis, R. (2005). Psychosocial group interventions for aged individuals suffering from Ioneliness. A randomized, controlled trial. *International Psychogeriatrics*, 17, 178.

Proudfoot 1999 {published data only}

Proudfoot, J., Gray, J., Carson, J., Guest, D., & Dunn, G. (1999). Psychological training improves mental health and job-finding among unemployed people. *International Archives of Occupational & Environmental Health*, 72(Suppl.), S40–S42.

Roth 2014 {published data only}

Roth, A. (2014). Engaging older cancer patients in psychotherapy and psychoeducation interventions: Care and vintage readers programs. *Psycho Oncology*, 23, 31.

Saito 2006 {published data only}

Saito, T. (2006). Development and evaluation of a social isolation prevention program for the relocated elderly. *Nippon Koshu Eisei Zasshi - Japanese Journal of Public Health*, 53(5), 338–346.

Savelkoul 2002 {published data only}

Savelkoul, M., Ritzen, W., & Konings, G. (2002). Social support and rheumatic diseases: A coping intervention for groups of patients. Gedrag & Gezondheid: Tijdschrift voor Psychologie en Gezondheid, 30(5), 351–357.

Shen 2012 {published data only}

Shen, P.-Y., Zhu, X.-Y., & Xin, Y. (2012). Effects of expressive group counseling on psychological well-being in children after earthquake. *Chinese Mental Health Journal*, 26(6), 466–470.

Shores 1998 {published data only}

Shores, M. M., Pascualy, M., & Veith, R. C. (1998). Major depression and heart disease: Treatment trials. Seminars in Clinical Neuropsychiatry, 3(2), 87–101.

Smith 2016 {published data only}

Smith, L. G., Banting, L., Eime, R., & van Uffelen, J. (2016). A systematic review of the literature concerning social support, loneliness, and

33 of 40

WILEY Campbell Collaboration

Star 1986 {published data only}

Star Toni, Z. (1986). Group social skills training: A comparison of two coaching programs. *Techniques*, 2(1), 24-38.

Tedder 1984 {published data only}

Tedder, S. L., Scherman, A., & Sheridan Kristin, M. (1984). Impact of group support on adjustment to divorce by single, custodial fathers. *American Mental Health Counselors Association Journal*, 6(4), 180–189.

Thalman 1994 {published data only}

Thalman, R. L (1994). A study of the effects of group support activities on spiritual well-being, social isolation, and anxiety of students in a personal health class [Masters dissertation, Utah State University]. https://elibrary.ru/item.asp?id=5657120

Titscher 2000 {published data only}

Titscher, G. (2000). Psychological factors and coronary heart disease. Journal fur Kardiologie, 7(6), 237–241.

Togher 2015 {published data only}

Togher, L. (2015). Addressing communication disorders for productive community integration. *Neuroepidemiology*, *45*(4), 303–304.

Tourigny 2005 {published data only}

Tourigny, M., Guillor, M.-L., & Morissette, P. (2005). Effectiveness of a group intervention for men sexually molested during childhood. Canadian Journal of Behavioural Science/Revue canadienne des sciences du comportement, 37(2), 97–109.

Van 2016 {published data only}

Van Orden, K. A., Swan, J. G., Parkhurst, K. A., Stone, D., Prothero, L. M., McIntosh, W., & Conwell, Y. (2016). Peer companionship for suicide risk reduction: the experience of older adults. *American Journal of Geriatric Psychiatry*, 24(3), S139–S140.

van den 2006 {published data only}

van den Elzen, A. J., & Fokkema, C. M. (2006). Home visits to the elderly in Leiden: an investigation into the effect on loneliness. *Tijdschrift Voor Gerontologie En Geriatrie*, 37(4), 142–146.

Victorson 2018 {published data only}

Victorson, D. (2018). Redefining the psychosocial adjustment of young adults with cancer through mindfulness training: Results from a randomized controlled pilot trial. *Psycho Oncology*, *27*, 51.

WELCH ET AL.

Xu 2002 {published data only}

Xu, L., Zhang, J., & Xiao, C. (2002). Psychotherapy for the clients with social intercourse difficulty. *Chinese Journal of Clinical Psychology*, 10(4), 287–289.

Yu 2021 {published data only}

Yu, D. (2021). Effect of cognitive behaviour therapy on the improvement of psychological stress. Work, 69(2), 515–529.

Zadeh {published data only}

Zadeh, B. M., Roustaei, A., Nadem, A. S., Asaadi, S., & Zadeh, N. G. Investigation of effectiveness of training resilience on rate aggression and loneliness of students. *Iranian Journal of Psychiatry*, 7(4 Suppl. 1), 41.

Zhang 2005 {published data only}

Zhang, W. H. (2005). Effects of group counseling on interpersonal relationship in poverty college students. *Chinese Journal of Clinical Rehabilitation*, 9(40), 13–15 (in Chinese).

Zhang et al., 2021 {published data only}

Zhang, Y., Xu, Y., Wang, H., Tang, J., Jiang, M., & Gu, Y. (2021). Effects of cognitive behavioral intervention on stigma and family support in patients with HIV infection/AIDS. *Chinese Nursing Research*, 35(16), 2869–2873. https://doi.org/10.12102/j.issn.1009-6493.2021.16.009

Zhou 2004 {published data only}

Zhou, Y. P., & Gao, X. C. (2004). Effect of psychodrama therapy on the trait-anxiety of patients with schizophrenia. Chinese Journal of Clinical Rehabilitation, 8(27), 5753–5755.

蒋宇星, 2021 {published data only}

蒋, 宇星, Yeates, C., & Bower Emily, S. 广场舞对于老年人孤独感及心理健 康的影响:一项范围综述. (2021). The impact of square dancing on loneliness and mental health in older adults: A scope review. In: 第二 十三届全国心理学学术会议. 中国内蒙古呼和浩特, 2.

강근모, 2015 {published data only}

강, 근모, & 이준기 (2015). Meta-analysis of effect of peer counseling program on adolescents. *Journal of Future Oriented Youth Society*, 12(3), 87-110.

김정숙, 2019 {published data only}

김, 정숙, 임영미, & 임정희 (2019). Effect of cognitive behavior group intervention in loneliness, depression, and suicidal ideation among the community-dwelling older adults with a high risk of depression. The Journal of Korean Nursing Research, 3(2), 25-35. https://doi.org/ 10.34089/jknr.2019.3.2.25

서경희 2010 {published data only}

서경희경희 경희 (2010). The effect of therapeutic music activities using the call & response singing form on the reduction of loneliness and the improvement of interpersonal relations of the senior citizen living at a nursing home. *Korean Journal of Music Therapy*, 12(3), 19–38.

위아름 2016 {published data only}

위아름아름 아름 (2016). The effect of music therapy influencing variables related with emotion of elderly: A meta-analysis. *Korean Journal of Music Therapy*, 18(2), 99-117.

ADDITIONAL REFERENCES

Abdi et al., 2019

Abdi, S., Spann, A., Borilovic, J., de Witte, L., & Hawley, M. (2019). Understanding the care and support needs of older people: A scoping review and categorisation using the WHO international classification of functioning, disability and health framework (ICF). BMC Geriatrics, 19(1), 1–15.

Akhter-Khan et al., 2020

Akhter-Khan, S. C., & Au, R. (2020). Why loneliness interventions are unsuccessful: A call for precision health. Advances in Geriatric Medicine and Research, 2(3):e200016. https://doi.org/10.20900/agmr20200016

Akhter-Khan et al., 2022

Akhter-Khan, S. C., Prina, M., Wong, G. H.-Y., Mayston, R., & Li, L. 2022). Understanding and addressing older adults' loneliness: The social relationship expectations framework. *Perspectives on Psychological Science*, 18(4), 762–777. https://doi.org/10.1177/ 17456916221127

Badcock et al., 2022

Badcock, J. C., Holt-Lunstad, J., Garcia, E., Bombaci, P., & Lim, M. H. (2022). Position statement: Addressing social isolation and loneliness and the power of human connection. Global Initiative on Loneliness and Connection. https://www.gilc.global/general-6

Bedney et al., 2010

Bedney, B. J., Goldberg, R. B., & Josephson, K. (2010). Aging in place in naturally occurring retirement communities: Transforming aging through supportive service programs. *Journal of Housing for the Elderly*, 24(3–4), 304–321.

Bee et al., 2014

Bee, P., Bower, P., Byford, S., Churchill, R., Calam, R., Stallard, P., Pryjmachuk, S., Berzins, K., Cary, M., & Wan, M. (2014). The clinical effectiveness, cost-effectiveness and acceptability of community-based interventions aimed at improving or maintaining quality of life in children of parents with serious mental illness: A systematic review. *Health Technology Assessment*, 18(8), 1–250.

Beller et al., 2018

Beller, J., & Wagner, A. (2018). Loneliness, social isolation, their synergistic interaction, and mortality. *Health Psychology*, 37(9), 808.

Campbell_WILEY-

Berkman et al., 2000

Berkman, L. F., Glass, T., Brissette, I., & Seeman, T. E. (2000). From social integration to health: Durkheim in the new millennium. *Social Science* & *Medicine*, 51(6), 843–857.

Bethell et al., 2019

Bethell, J., Babineau, J., Iaboni, A., Green, R., Cuaresma-Canlas, R., Karunananthan, R., Schon, B., Schon, D., & McGilton, K. S. (2019). Social integration and loneliness among long-term care home residents: Protocol for a scoping review. *BMJ Open*, 9(12), e033240.

Boulton et al., 2021

Boulton, E., Kneale, D., Stansfield, C., Heron Paul, N., Sutcliffe, K., Hayanga, B., Hall, A., Bower, P., Casey, D., & Craig, D. (2021). Rapid systematic review of systematic reviews: What befriending, social support and low intensity psychosocial interventions, delivered remotely, may reduce social isolation and loneliness among older adults and how? [version 2; peer review: 2 approved with reservations]. *F1000Research*, *9*, 1368.

Cattan et al., 2005

Cattan, M., White, M., Bond, J., & Learmouth, A. (2005). Preventing social isolation and loneliness among older people: A systematic review of health promotion interventions. *Ageing & Society*, 25(1), 41–67.

CDC 2015

Center for Disease Control. (2015). The social-ecological model: A *framework for prevention*. National Center for Injury Prevention and Control, Division of Violence Prevention.

Cené et al., 2022

Cené, C. W., Beckie, T. M., Sims, M., Suglia, S. F., Aggarwal, B., Moise, N., Jiménez, M. C., Gaye, B., & McCullough, L. D. (2022). Effects of objective and perceived social isolation on cardiovascular and brain health: A scientific statement from the American Heart Association. *Journal of the American Heart Association*, 11(16), e026493.

Cohen-Mansfield et al., 2015

Cohen-Mansfield, J., & Perach, R. (2015). Interventions for alleviating loneliness among older persons: A critical review. American Journal of Health Promotion, 29(3), e109–e125.

Cotterell et al., 2018

Cotterell, N., Buffel, T., & Phillipson, C. (2018). Preventing social isolation in older people. *Maturitas*, 113, 80–84.

Dahlberg et al., 2022

Dahlberg, L., McKee, K. J., Frank, A., & Naseer, M. (2022). A systematic review of longitudinal risk factors for loneliness in older adults. *Aging & Mental Health*, 26(2), 225–249.

Dassieu et al., 2021

Dassieu, L., & Sourial, N. (2021). Tailoring interventions for social isolation among older persons during the COVID-19 pandemic: challenges and pathways to healthcare equity. *International Journal for Equity in Health*, 20(1), 1–4.

Dickens et al., 2011

Dickens, A. P., Richards, S. H., Greaves, C. J., & Campbell, J. L. (2011). Interventions targeting social isolation in older people: A systematic review. BMC Public Health, 11(1), 1–22.

Digital Solution Foundry and EPPI_Centre 2020

Digital Solution Foundry and EPPI_Centre. (2020). *Eppi-Mapper*;(1.2.5) [Computer software].

Dodge 2020

Dodge, K. A. (2020). Annual Research Review: Universal and targeted strategies for assigning interventions to achieve population impact. *Journal of Child Psychology and Psychiatry*, 61(3), 255–267. https:// doi.org/10.1111/jcpp.13141

Donovan et al., 2020

Donovan Nancy, J., & Dan, B. (2020). Social isolation and loneliness in older adults: review and commentary of a National Academies report. The American Journal of Geriatric Psychiatry, 28(12), 1233–1244.

Eccles et al., 2021

Eccles, A. M., & Pamela, Q. (2021). Alleviating loneliness in young people– A meta-analysis of interventions. *Child and Adolescent Mental Health*, 26(1), 17–33.

Elder et al., 2012

Elder, K., & Retrum, J. (2012). Framework for isolation in adults over 50. AARP Foundation.

Ernst et al., 2022

Ernst, M., Niederer, D., Werner, A. M., Czaja, S. J., Mikton, C., Ong, A. D., Rosen, T., Brähler, E., & Beutel, M. E. (2022). Loneliness before and during the COVID-19 pandemic: A systematic review with metaanalysis. *American Psychologist*, *77*(5), 600–677.

Fakoya et al., 2020

Fakoya, O. A., McCorry, N. K., & Donnelly, M. (2020). Loneliness and social isolation interventions for older adults: A scoping review of reviews. BMC Public Health, 20(1), 1–14.

Findlay 2003

Findlay, R. A. (2003). Interventions to reduce social isolation amongst older people: Where is the evidence? Ageing & Society, 23(5), 647–658.

Galvez-Hernandez et al., 2022

Galvez-Hernandez, P., González-de Paz, L., & Muntaner, C. (2022). Primary care-based interventions addressing social isolation and loneliness in older people: A scoping review. *BMJ Open*, *12*(2), e057729.

Gardiner et al., 2018

Gardiner, C., Geldenhuys, G., & Gott, M. (2018). Interventions to reduce social isolation and loneliness among older people: An integrative review. *Health & Social Care in the Community*, *26*(2), 147–157.

Grenade et al., 2008

Grenade, L., & Boldy, D. (2008). Social isolation and loneliness among older people: Issues and future challenges in community and residential settings. Australian Health Review, 32(3), 468–478.

Hagan et al., 2014

Hagan, R., Manktelow, R., Taylor, B. J., & Mallett, J. (2014). Reducing loneliness amongst older people: A systematic search and narrative review. Aging & Mental Health, 18(6), 683–693.

Holt-Lunstad 2018

Holt-Lunstad, J. (2018). Why social relationships are important for physical health: A systems approach to understanding and modifying risk and protection. *Annual Review of Psychology*, *69*, 437–458.

Holt-Lunstad 2022

Holt-Lunstad, J. (2022). Social connection as a public health issue: The evidence and a systemic framework for prioritizing the "social" in social determinants of health. *Annual Review of Public Health*, 43, 193–213.

Holt-Lunstad et al., 2015

Holt-Lunstad, J., Smith, T. B., Baker, M., Harris, T., & Stephenson, D. (2015). Loneliness and social isolation as risk factors for mortality: A metaanalytic review. *Perspectives on Psychological Science*, 10(2), 227–237.

Holwerda et al. 2014

Holwerda, T. J., Deeg, D. J., Beekman, A. T., Van Tilburg, T. G., Stek, M. L., Jonker, C., & Schoevers, R. A. (2014). Feelings of loneliness, but not social isolation, predict dementia onset: Results from the Amsterdam Study of the Elderly (AMS^{TEL}). *Journal of Neurology, Neurosurgery & Psychiatry*, *85*(2), 135–142.

Jarvis et al., 2020

Jarvis, M.-A., Padmanabhanunni, A., Balakrishna, Y., & Chipps, J. (2020). The effectiveness of interventions addressing loneliness in older persons: an umbrella review. *International Journal of Africa Nursing Sciences*, 12, 100177.

Jopling 2020

Jopling, K. (2020). Promising approaches revisited: Effective action on loneliness in later life. Campaign to End Loneliness.

Kelly et al., 2017

Kelly, M. E., Duff, H., Kelly, S., Power, J. E. M., Brennan, S., Lawlor, B. A., & Loughrey, D. G. (2017). The impact of social activities, social networks, social support and social relationships on the cognitive functioning of healthy older adults: A systematic review. Systematic Reviews, 6(1), 1–18.

Lasgaard et al., 2016

Lasgaard, M., Friis, K., & Shevlin, M. (2016). "Where are all the lonely people?" A population-based study of high-risk groups across the life span. Social Psychiatry and Psychiatric Epidemiology, 51, 1373–1384.

Leigh-Hunt et al., 2017

Leigh-Hunt, N., Bagguley, D., Bash, K., Turner, V., Turnbull, S., Valtorta, N., & Caan, W. (2017). An overview of systematic reviews on the public health consequences of social isolation and loneliness. *Public Health*, 152, 157–171.

Lim et al., 2020

Lim, M. H., Eres, R., & Vasan, S. (2020). Understanding loneliness in the twenty-first century: an update on correlates, risk factors, and potential solutions. *Social Psychiatry and Psychiatric Epidemiology*, 55(7), 793–810.

Mann et al., 2017

Mann, F., Bone, J. K., Lloyd-Evans, B., Frerichs, J., Pinfold, V., Ma, R., Wang, J., & Johnson, S. (2017). A life less lonely: The state of the art in interventions to reduce loneliness in people with mental health problems. *Social Psychiatry and Psychiatric Epidemiology*, 52(6), 627–638.

Masi et al., 2011

Masi, C. M., Chen, H.-Y., Hawkley, L. C., & Cacioppo John, T. (2011). A meta-analysis of interventions to reduce loneliness. *Personality and* social psychology review, 15(3), 219–266.

37 of 40

McGowan et al., 2016

McGowan, J., Sampson, M., Salzwedel Douglas, M., Cogo, E., Foerster, V., & Lefebvre, C. (2016). PRESS peer review of electronic search strategies: 2015 guideline statement. *Journal of Clinical Epidemiology*, 75, 40–46.

Menec et al., 2020

Menec, V. H., Newall, N. E., Mackenzie, C. S., Shooshtari, S., & Nowicki, S. (2020). Examining social isolation and loneliness in combination in relation to social support and psychological distress using Canadian Longitudinal Study of Aging (CLSA) data. *PLoS One*, 15(3), e0230673.

Moher et al., 2015

Moher, D., Shamseer, L., Clarke, M., Ghersi, D., Liberati, A., Petticrew, M., Shekelle, P., & Stewart, L. A. (2015). Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015 statement. Systematic Reviews, 4(1), 1–9.

Moore et al., 2018

Moore, T. H. M., Kesten, J. M., López-López, J. A., Ijaz, S., McAleenan, A., Richards, A., & Audrey, S. (2018). The effects of changes to the built environment on the mental health and well-being of adults: Systematic review. *Health & Place*, *53*, 237–257.

NASEM 2020

NASEM. (2020). Social isolation and loneliness in older adults: Opportunities for the health care system. National Academies Press.

O'Neill et al., 2014

O'Neill, J., Tabish, H., Welch, V., Petticrew, M., Pottie, K., Clarke, M., Evans, T., Pardo, J. P., Waters, E., & White, H. (2014). Applying an equity lens to interventions: Using PROGRESS ensures consideration of socially stratifying factors to illuminate inequities in health. *Journal* of Clinical Epidemiology, 67(1), 56–64.

Ogrin et al., 2021

Ogrin, R., Cyarto Elizabeth, V., Harrington Karra, D., Haslam, C., Lim Michelle, H., Golenko, X., Bush, M., Vadasz, D., Johnstone, G., & Lowthian Judy, A. (2021). Loneliness in older age: What is it, why is it happening and what should we do about it in Australia? *Australasian Journal on Ageing*, 40(2), 202–207.

O'Rourke et al. 2018

O'Rourke, H. M., Collins, L., & Sidani, S. (2018). Interventions to address social connectedness and loneliness for older adults: A scoping review. *BMC Geriatrics*, 18(1), 214.

O'Sullivan et al., 2021

O'sullivan, R., Burns, A., Leavey, G., Leroi, I., Burholt, V., Lubben, J., Holt-Lunstad, J., Victor, C., Lawlor, B., & Vilar-Compte, M. (2021). Impact

WILEY-Campbell Collaboration

of the COVID-19 pandemic on loneliness and social isolation: A multi-country study. International Journal of Environmental Research and Public Health, 18(19), 9982.

Perlman et al., 1981

Perlman, D., & Peplau, L. A. (1981). Toward a social psychology of loneliness. *Personal Relationships*, 3, 31–56.

Poscia et al., 2018

Poscia, A., Stojanovic, J., La Milia, D. I., Duplaga, M., Grysztar, M., Moscato, U., Onder, G., Collamati, A., Ricciardi, W., & Magnavita, N. (2018). Interventions targeting loneliness and social isolation among the older people: An update systematic review. *Experimental Gerontology*, 102, 133–144.

Proffitt et al., 1993

Proffitt, C., & Byrne, M. (1993). Predicting loneliness in the hospitalized elderly: What are the risk factors? *Geriatric Nursing*, 14(6), 311–314.

Prohaska et al., 2020

Prohaska, T., Burholt, V., Burns, A., Golden, J., Hawkley, L., Lawlor, B., Leavey, G., Lubben, J., O'Sullivan, R., & Perissinotto, C. (2020). Consensus statement: Loneliness in older adults, the 21st century social determinant of health? *BMJ Open*, 10(8), e034967.

Qualter et al., 2015

Qualter, P., Vanhalst, J., Harris, R., Van Roekel, E., Lodder, G., Bangee, M., Maes, M., & Verhagen, M. (2015). Loneliness across the life span. *Perspectives on Psychological Science*, 10(2), 250–264. https://doi. org/10.1177/1745691615568999

Qualter et al., 2022

Qualter, P., Eccles, A., & Barreto, M. (2022). Chapter 8. Interventions for loneliness in younger people. In D. V. Jeste, T. T. Nguyen, & N. J. Donovan (Eds.), *Loneliness: Science and practice* (1st ed.), pp. 165–202. American Psychiatric Publishing.

Rafnsson et al. 2020

Rafnsson, S. B., Orrell, M., d'Orsi, E., Hogervorst, E., & Steptoe, A. (2020). Loneliness, social integration, and incident dementia over 6 years: Prospective findings from the English longitudinal study of ageing. *The Journals of Gerontology: Series B*, 75(1), 114-124.

Razai et al., 2020

Razai, M. S., Oakeshott, P., Kankam, H., Galea, S., & Stokes-Lampard, H. (2020). Mitigating the psychological effects of social isolation during the COVID-19 pandemic. *BMJ*, 369, m1904.

Reichow et al., 2012

Reichow, B., Steiner, A. M., & Volkmar, F. (2012). Social skills groups for people aged 6 to 21 with autism spectrum disorders (ASD). *Campbell Systematic Reviews*, 8(1), 1–76.

Röhr et al., 2021

Röhr, S., Wittmann, F., Engel, C., Enzenbach, C., Witte, A. V., Villringer, A., Löffler, M., & Riedel-Heller, S. G. (2021). Social factors and the prevalence of social isolation in a population-based adult cohort. *Social Psychiatry and Psychiatric Epidemiology*, *57*(10), 1959–1968. https://doi.org/10.1007/s00127-021-02174-x

Shea et al., 2017

Shea, B. J., Reeves, B. C., Wells, G., Thuku, M., Hamel, C., Moran, J., Moher, D., Tugwell, P., Welch, V., & Kristjansson, E. (2017). AMSTAR
2: A critical appraisal tool for systematic reviews that include randomised or non-randomised studies of healthcare interventions, or both. *BMJ*, 358, j4008.

Snilstveit 2016

Snilstveit, B., Vojtkova, M., Bhavsar, A., Stevenson, J., & Gaarder, M. (2016). Evidence & Gap Maps: A tool for promoting evidence informed policy and strategic research agendas. *Journal of Clinical Epidemiology*, 79, 120-129.

Snilstveit et al., 2013

Snilstveit, B., Vojtkova, M., Bhavsar, A., & Gaarder, M. (2013). Evidence gap maps—A tool for promoting evidence-informed policy and prioritizing future research. The World Bank.

Springer et al., 2007

Springer, J. F., & Phillips, J. (2007). The Institute of Medicine Framework and its implication for the advancement of prevention policy, programs and practice. Institute of Medicine of the National Academies.

Surkalim et al., 2022

Surkalim, D. L., Luo, M., Eres, R., Gebel, K., van Buskirk, J., Bauman, A., & Ding, D. (2022). The prevalence of loneliness across 113 countries: Systematic review and meta-analysis. *BMJ*, 376, e067068.

Teo et al., 2022

Teo, R. H., Cheng, W. H., Cheng, L. J., Ying, L. A. U., & Tiang, S. (2022). Global prevalence of social isolation among community-dwelling older adults: A systematic review and meta-analysis. Archives of Gerontology and Geriatrics, 107, 104904. https://doi.org/10.1016/j.archger.2022.104904

Thomas et al., 2020

Thomas, J., Graziosi, S., Brunton, J., Ghouze, Z., O'Driscoll, P., & Bond, M. (2020). EPPI-Reviewer: Advanced software for systematic reviews, maps and evidence synthesis. Eppi-Centre Software. UCL Social Research Institute.

Valtorta et al., 2016

Valtorta, N. K., Kanaan, M., Gilbody, S., Ronzi, S., & Hanratty, B. (2016). Loneliness and social isolation as risk factors for coronary heart disease and stroke: Systematic review and meta-analysis of longitudinal observational studies. *Heart*, 102(13), 1009–1016.

Veronese et al., 2021

Veronese, N., Galvano, D., D'Antiga, F., Vecchiato, C., Furegon, E., Allocco, R., Smith, L., Gelmini, G., Gareri, P., & Solmi, M. (2021). Interventions for reducing loneliness: An umbrella review of intervention studies. *Health & Social Care in the Community*, 29(5), e89–e96.

Victor et al., 2018

Victor, C., Mansfield, L., Kay, T., Daykin, N., Lane, J., Duffy Lily, G., Tomlinson, A., & Meads, C (2018). What works centre for wellbeing. https://whatworkswellbeing.org/wp/wp-content/uploads/ woocommerce_uploads//10/Full-report-Tackling-loneliness-Oct-2018.pdf

Waddington et al., 2014

Waddington, H., Snilstveit, B., Hombrados, J., Vojtkova, M., Phillips, D., Davies, P., & White, H. (2014). Farmer field schools for improving farming practices and farmer outcomes: A systematic review. *Campbell Systematic Reviews*, 10(1), i–335.

Welch et al., 2022

Welch, V., Ghogomu, E. T., Barbeau, V. I., Boulton, E., Boutin, S., Haitas, N., Kneale, D., Salzwedel, D. M., Simard, R., Herbert, P., & Mikton, C. (2022). Digital interventions to reduce social isolation and loneliness in older adults: An evidence and gap map. *Campbell Systematic Reviews*, 18(3), e1260. https://doi.org/10.1002/cl2.1260

Welch et al., 2022b

Welch, V., Ghogomu, E., Barbeau, V., Dowling, S., Doyle, R., Beveridge, E., Boulton, E., Desai, P., Huang, J., Elmestekawy, N., Hussain, T., Wadhwani, A., Boutin, S., Haitas, N., Kneale, D., Salzwedel, D. M., Simard, R., Hebert, P., & Mikton, C. (2022). Digital interventions to reduce social isolation and loneliness in older adults: An evidence and gap map. SocArXiv. https://doi.org/10.31235/osf.io/736rp

Welch et al., 2023

Welch, V., Tanjong Ghogomu, E., Dowling, S., Choo, W. Y., Yunus, R. M., Mohd, T. A. T., & Mikton, C. (2023). PROTOCOL: In-person interventions to reduce social isolation and loneliness: An evidence and gap map. *Campbell Systematic Reviews*, 19(3), e1340.

White et al., 2020

White, H., Albers, B., Gaarder, M., Kornør, H., Littell, J., Marshall, Z., Matthew, C., Pigott, T., Snilstveit, B., & Waddington, H. (2020). Campbell_WILEY-

Guidance for producing a Campbell evidence and gap map. *Campbell Systematic Reviews*, 16(4), e1125.

WHO 2002

WHO. (2002). World report on violence and health.

WHO 2019

World Health Organization. (2019). World health statistics: Monitoring health for the SDGs, sustainable development goals.

WHO 2021

World Health Organization. (2021). Social isolation and loneliness among older people: advocacy brief.

Williams et al., 2021

Williams Christopher, Y. K., Townson Adam, T., Kapur, M., Ferreira Alice, F., Nunn, R., Galante, J., Phillips, V., Gentry, S., & Usher-Smith, J. A. (2021). Interventions to reduce social isolation and loneliness during COVID-19 physical distancing measures: A rapid systematic review. *PLoS One*, 16(2), e0247139.

Windle et al., 2012

Windle, K., Francis, J., & Coomber, C. (2012). Preventing loneliness and social isolation: Interventions and outcomes (SCIE Research Briefing 39). Social Care Institute for Excellence.

Wold et al., 2018

Wold, B., & Mittelmark, M. B. (2018). Health-promotion research over three decades: The social-ecological model and challenges in implementation of interventions. *Scandinavian Journal of Public Health*, 46(20-Suppl.), 20-26.

World Bank 2022

World Bank. (2022). World Bank country and lending groups. Country classification. World Bank Data Help Desk.

Yang et al., 2011

Yang, K., & Victor, C. (2011). Age and loneliness in 25 European nations. Ageing & Society, 31(8), 1368–1388.

Zamir et al., 2018

Zamir, S., Hennessy, C. H., Taylor, A. H., & Jones, R. B. (2018). Video-calls to reduce loneliness and social isolation within care environments for older people: An implementation study using collaborative action research. BMC Geriatrics, 18(1), 1–13.

Zavaleta et al., 2014

Zavaleta, D., Samuel, K., & Mills, C. (2014). Social isolation: A conceptual and measurement proposal (OPHI Working Paper No. 67). Oxford Poverty & Human Development Initiative, University of Oxford.

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article.

How to cite this article: Welch, V., Ghogomu, E. T., Dowling, S., Barbeau, V. I., Al-Zubaidi, A. A. A., Beveridge, E., Bondok, M., Desai, P., Doyle, R., Huang, J., Hussain, T., Jearvis, A., Jahel, F., Madani, L., Choo, W. Y., Yunus, R. M., Tengku Mohd, T. A. M., Wadhwani, A., Ameer, A. A., ... Mikton, C. (2024). In-person interventions to reduce social isolation and loneliness: An evidence and gap map. *Campbell Systematic Reviews*, 20, e1408. https://doi.org/10.1002/cl2.1408